

THE SCHOOL BOARD OF MIAMI-DADE COUNTY, FLORIDA SCHOOL BOARD ADMINISTRATION BUILDING

Procurement Management Services 1450 N.E. 2nd Avenue, Room 650 Miami, FL 33132

PUBL	de schools	Direct All Inquiries To Procurement Management Services	
		Buyer's Name:	
		PHONE: (305) 995	
		Email:	
	BID/RFP ADDENDUM	` '	
		Date:	
		Addendum No.	
BID/F	RFP NoBID/RFP TITLE:		
This	addendum modifies the conditions of the above-	referenced BID/RFP as follows:	
	formation, specifications terms, and conditions for the ed on the Procurement Management website at http:/	e above-referenced BID/RFP, are included on the document	
refere your	enced bid. If your bid/proposal has not been submentire bid/proposal package.	formation and requirements constitute an integral part of the itted, substitute the pages marked REVISED and mail	
I ack	nowledge receipt of Addendum Number		
	ASE NOTE: If your firm has forwarded a copy of thi ard him/her a copy of this addendum.	is bid/proposal to another vendor, it is your responsibility to	
	(PLEASE TYPE	OR PRINT BELOW)	
LEGA	L NAME OF BIDDER:		
MAILI	NG ADDRESS:		
CITY,	STATE ZIP CODE:		
TELEI	PHONE NUMBER: E-MAIL	FAX #	
BY:	SIGNATURE (Manual):		
	OF AUTHORIZED REPRESENTATIVE		
	NAME (Typed):	TITLE:	

OF AUTHORIZED REPRESENTATIVE

GENERAL CONDITIONS OF THE CONTRACT FOR CONSTRUCTION

TABLE OF ARTICLES

1.	GENERAL PROVISIONS
1.1 1.2 1.3 1.4 1.5	Basic Definitions Execution, Correlation and Intent Ownership and Use of Documents Capitalization Interpretation
2.	BOARD
2.1 2.2 2.3 2.4 2.5	Definition Information and Services Required of Board Board's Right to Stop the Work Board's Right to Carry Out the Work Board's Right to Require Documentation & Audit
3.	CONTRACTOR
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9 3.10	Definition Review of Construction Documents and Field Conditions by Contractor Supervision and Construction Procedures Labor, Materials, Products and Equipment Substitutions Warranty Taxes Permits, Fees, Notices, Jessica Lunsford Act and K-20 School Code Superintendent Contractor's Project Schedule
3.11 3.12	Documents and Samples at the Site Shop Drawings, Product Data and Samples
3.13 3.14 3.15 3.16	Use of Site Cutting and Patching Cleaning Up and Salvage Access to Work
3.17 3.18	Royalties and Patents Indemnification

Table of Contents for General Conditions of the Contract for Construction – Rev. 12-5-2011

	3.19 3.20 3.21	Testing Energy Efficient and Sustainable Buildings Construction Site Compliance Monitoring
4.		ADMINISTRATION OF THE CONTRACT
	4.1 4.2 4.3 4.4	Architect or Engineer (A/E) A/E's Administration of the Contract Claims and Disputes Resolution of Claims and Disputes
5.		SUBCONTRACTORS
	5.1 5.2 5.3 5.4	Definitions Award of Subcontracts and Other Contracts for Portions of the Work Subcontractual Relations Contingent Assignment of Subcontracts
6.		CONSTRUCTION BY BOARD OR BY SEPARATE CONTRACTORS
	6.1 6.2 6.3	Board's Right to Perform Construction and to Award Separate Contracts Mutual Responsibility Board's Right to Clean Up
7.		CHANGES IN THE WORK
	7.1 7.2 7.3	Change Orders Reserved for Future Use Field Orders
8.		TIME
	8.1 8.2 8.3	Time Progress and Completion Delays and Extensions of Time
9.		PAYMENTS AND COMPLETION
	9.1 9.2 9.3 9.4 9.5 9.6 9.7 9.8	The Contract Sum Schedule of Values Requisitions for Payment Certificates for Payment Decisions to Withhold Certification Progress Payments Timely Payment Substantial Completion

9.9 9.10 9.11	Partial Occupancy or Use Final Completion and Final Payment Prompt Payment Act
10.	PROTECTION OF PERSONS AND PROPERTY
10.1 10.2 10.3	Safety Precautions and Programs Safety of Persons and Property Emergencies
11.	INSURANCE AND BONDS
11.1 11.2 11.3	Contractor's Insurance Board Provided Property Insurance Program Bonds
12.	UNCOVERING AND CORRECTION OF WORK
12.1 12.2 12.3	Uncovering of Work Correction of Work, Guarantee and Warranty Acceptance of Nonconforming Work
13.	MISCELLANEOUS PROVISIONS
13.1 13.2 13.3 13.4 13.5 13.6 13.7	Governing Law and Venue Successors and Assigns Written Notice Rights and Remedies Tests and Inspections Commencement of Statutory Limitation Period Provision of Site Amenities
14.	TERMINATION OR SUSPENSION OF THE CONTRACT
14.1 14.2 14.3 14.4	Termination by Contractor Termination by Board for Cause Suspension by Board for Convenience Termination by Board for Convenience
15.	WATER AND SEWER
16.	WATER AND SEWER (CONSTRUCTION MANAGER AT-RISK only)
17.	FORMS

Table of Contents for General Conditions of the Contract for Construction – Rev. 12-5-2011

GENERAL CONDITIONS OF THE CONTRACT FOR CONSTRUCTION

ARTICLE 1 - GENERAL PROVISIONS

1.1 BASIC DEFINITIONS

1.1.1 The Construction Documents

The Construction Documents consist of the Legal Advertisement Covering Opening of Bids, Instructions to Bidders, the Public Construction Payment Bond, the Public Construction Performance and Guarantee Bond, if required, the Trench Safety Compliance Statement, Special Provisions for Compliance with M/WBE Subcontracting Assistance Levels (if applicable), Conditions of the Contract (general, supplementary, special and other Conditions), the Plans, the Specifications and Addenda, the Bid and Accepted Alternates, List of Subcontractors, Licensure Certification, Certificate of Insurance, all incorporated in the Contract before its execution, together with this Agreement and all Modifications authorized pursuant to Article 7, changes in the work, herein.

1.1.2 The Contract

The Construction Documents form the Contract for construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either oral or written. The Contract may be amended or modified only by a Modification authorized pursuant to Article 7, changes in the work, herein. The Construction Documents shall not be construed to create a contractual relationship of any kind (1) between A/E and Contractor, (2) between Board and a Subcontractor or Sub-subcontractor, or (3) between any persons or entities other than Board and Contractor.

1.1.3 The Work

The term "Work" means the construction and services required by the Contract and Construction Documents, whether completed or partially completed, and reasonably inferable therefrom and that would have been reasonably observable and reasonably inferable from a contractor's inspection of the site, and includes all labor, materials, products, equipment and services provided or to be provided by the Contractor to fulfill its obligations. The Work may constitute the whole or a part of the Project.

1.1.4 The Project

The Project is the total construction of which the Work performed under the Construction Documents may be the whole or a part and which may include construction by Board or by separate contractors.

1.1.5 The Drawings

The Drawings are the graphic and pictorial portions of the Construction Documents, wherever located and wherever issued, schematically showing the design, location, and dimensions of the Work, including plans, elevations, sections, details, schedules and diagrams. The Drawings are intended to show general arrangements, design and extent of Work, and are not intended to serve as Shop Drawings.

1.1.6 The Specifications

The Specifications are that portion of the Construction Documents consisting of the written requirements for materials, equipment, construction systems, standards and workmanship for the Work and performance of related services. Specifications are separated into titled divisions for convenience of reference only, and shall not be interpreted as establishing divisions for the Work, trades, Subcontracts or extent of any part of the Work. The Specifications is the volume(s) usually assembled for the Work which may include the Instructions to Bidders, Bidding Requirements, Sample forms, Conditions of Contract and Specifications.

1.1.7 Document Control

Document Control as used herein shall mean the Board's Representative for the purpose of being the custodian of all plans, documents, correspondence and drawings for the project.

1.1.8 Acceptance

As used herein Acceptance shall mean that all of the work required by the contract (or by the individual work order issued under the contract) is fully executed and completed in accordance with the plans and specifications so that no work remains to be completed. This shall require and include that all close-out documentation be fully completed, submitted, and approved.

1.1.9. Final Completion

Final Completion shall have the same definition and be used interchangeably with Acceptance.

1.1.10 Chief Building Official and Building Code Consultant (BCC)

The officer or other designated authority, or their duly authorized representative charged with the administration and enforcement of the Florida Building Code (FBC) which for purposes of construction on Board owned property is that individual so designated by the Board as the Board's Chief Building Official in charge of the Board's Office of Educational Facilities Compliance. The Building Code Consultant is the firm retained by the Board to perform plan reviews and inspections of the work for compliance with the Florida Building Code, unless such plan reviews and/or inspections are performed by the Board's employees.

1.1.11 Local Governmental Review

The Local Government Review ("LGR") refers to the review of Capital Improvement Projects to ensure consistency with local comprehensive master developmental plans and local land development regulations.

1.2 **EXECUTION, CORRELATION AND INTENT**

- 1.2.1 The Contract is binding on the date of Award by Board. Before commencing Work, Contractor shall furnish the following: (a) Contract Bond(s) as provided in Article 11.1 evidence of insurance as provided in Article 11.3, a signed Lump Sum Contract form and Licensure Certification form. Failure of Contractor to file with Board an acceptable bond and insurance as provided for in Article 11 within twenty (20) calendar days from the date of the Award shall be considered just cause and Board may annul and void the Award and make demand upon and declare forfeiture of the bid bond, proposal guarantee or a good faith deposit in liquidation of all damages sustained.
- 1.2.2 Failure of Contractor to execute the Contract within forty-five (45) calendar days of the Award may be just cause for Board to annul and void the Award. Award may then be made by Board to the next lowest responsible Bidder or the Work may be re-advertised or may be constructed by day labor, or any other means necessary at Board's discretion or as permitted by applicable law.
- 1.2.3 Execution of the Contract by Contractor is a representation that Contractor has visited

the site, become familiar with local conditions under which the Work is to be performed, correlated personal observations with requirements of the Construction Documents, and reported to Board any error, inconsistency or omission, or variance with applicable laws, codes and regulations.

- 1.2.4 The intent of the Construction Documents is to include all items necessary for the proper execution and completion of the Work by Contractor including all labor and materials, products, equipment, implements, machinery, tools, storage, apparatus and means of transportation necessary for the proper execution of the Work. The Construction Documents are complementary and what is required by one shall be as binding as if required by all; performance by a Contractor shall be required to the extent consistent with the Construction Documents, and reasonably inferable therefrom and that would have been reasonably observable and reasonably inferable from a contractor's pre-bid inspection of the site, as being necessary to produce the intended results.
- 1.2.5 Reference in the Specifications to manufacturers' specifications and instructions means the latest edition in effect at the date of bidding.
- 1.2.6 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings will not control Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade. No responsibility, direct or implied, is assumed by Board or A/E for omission or duplication by Contractor or Subcontractors due to real or alleged error in organization of the Construction Documents.
- 1.2.7 Unless otherwise stated in the Construction Documents, words which have well-known technical or construction industry meanings are used in the Construction Documents in accordance with such recognized meanings.

1.3 OWNERSHIP AND USE OF DOCUMENTS

1.3.1. The Drawings, Specifications and other documents furnished to Contractor are and shall remain Board's property. All copies, except the Contractor's record set which may be retained, shall be returned if possible or suitably accounted for upon completion of the Work. Neither Contractor nor any Subcontractor, Sub-subcontractor or material or equipment supplier shall own or claim a copyright on the drawings, specifications, schedules and other documents prepared in connection with the Work. The Contract Documents furnished to or by Contractor for use solely with respect to this Project are not to be used by Contractor, any Subcontractor, Sub-subcontractor or material or equipment supplier on other Projects or for additions to this Project

outside the scope of the Work without Board's specific written consent.

1.4 **CAPITALIZATION**

1.4.1 Terms capitalized in these General Conditions include but are not limited to those which are (1) specifically defined, or (2) titles of numbered articles and identified references to Paragraphs, Subparagraphs and Clauses in the Construction Documents.

1.5 **INTERPRETATION**

- 1.5.1 .1 In the interest of brevity, the Construction Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.
 - .2 It is the intent of the Contract Documents to describe a functionally complete Project (or part thereof) to be constructed in accordance with the Contract Documents. Any Work, materials, products or equipment that may reasonably be inferred from the Contract Documents and that would have been reasonably observable and reasonably inferable from a contractor's inspection of the site, as being required to produce the intended result will be supplied whether or not specifically called for.
 - .3 Where there is a discrepancy, Supplementary and Special Conditions are intended to modify and take precedence over General Conditions. In the event of a discrepancy between the drawings and specifications, the specifications shall be followed. Anything shown in one and not the other, and anything obviously necessary to complete the Project and achieve the intended result although not shown or described in the one or the other, shall be brought to the attention of the A/E, and if so directed by the A/E shall be provided or performed by the Contractor as part of its Contract.

ARTICLE 2 - BOARD

2.1 **DEFINITION**

2.1.1 Board refers to the School Board of Miami-Dade County, Florida or any of its authorized representatives.

SECTION 00700

2.2 INFORMATION AND SERVICES REQUIRED OF BOARD

- 2.2.1 Board will furnish surveys in the Specifications. Board may make available upon request soil borings for information only but such are not to be relied on by the contractor.
- 2.2.2 All easements and rights-of-way will be procured and paid for by Board unless otherwise specifically provided in the Construction Documents.
- 2.2.3 Information or services under Board's control shall be furnished by Board with reasonable promptness to avoid delay in orderly progress of the Work.
- 2.2.4 Unless otherwise provided in the Construction Documents, Contractor will be furnished free of charge six (6) copies of drawings and the Specifications.
- 2.2.5 The foregoing are in addition to other duties and responsibilities of Board enumerated herein and especially in respect to Article 6 (Construction by Board or by Separate Contractors), Article 9 (Payments and Completion) and Article 11 (Insurance and Bonds).

2.3 **BOARD'S RIGHT TO STOP THE WORK**

2.3.1 If Contractor fails to correct Work which is not in accordance with the requirements of the Construction Documents as required by Paragraph 12.2, or fails to carry out Work in accordance with the Construction Documents or at the A/E's direction, Board may, in writing, order Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of Board to stop the Work shall not give rise to a duty on the part of Board to exercise this right for the benefit of Contractor or any other person or entity, except to the extent required by Subparagraph 6.1.3. Such order, properly given due to contractor's failure as defined above, shall not create the right for the contractor to receive a time extension or any excusable or compensable delay.

2.4 BOARD'S RIGHT TO CARRY OUT THE WORK

2.4.1 If Contractor defaults or neglects to carry out the Work properly and diligently in substantial accord and compliance with the Project Schedule(s) furnished to and acceptable to A/E, or if Contractor fails or refuses to carry out the Work in accordance with the Construction Documents, Board reserves the right to notify Contractor, in writing, listing the specific items to be performed and the time in which performance is to be accomplished and, if not performed or proceeded with within the

General Conditions of the Contract for Construction - Rev. 12-5-2011

time specified, Board may, without prejudice to other remedies Board may have, take over the Work or such portion thereof as may be in default, and correct and make good the deficiencies. In such case, the cost thereof, including compensation for A/E's additional services and expenses made necessary by such default, neglect or failure, may be deducted from any amount due or to become due Contractor from Board. If payments then or thereafter due Contractor are not sufficient to cover such amounts, Contractor or surety shall pay the difference to Board.

2.5 BOARD'S RIGHT TO REQUIRE DOCUMENTATION & AUDIT

Board may, as it deems necessary, require from Contractor support and/or documentation for any submission made by Contractor to A/E and/or Board.

Board shall have access, and contractor upon execution of the contract gives the Board unrestricted access during normal working hours, to all Contractor's records relating to this project including hard copy as well as electronic records for a period of two years after project acceptance. In the event such audit reveals any discrepancies in quantities of services or level of compensation from the provisions of the Contract, the Board shall have the right to recover costs, including auditor's salary plus fringe benefits as a result of overcharge by the Contractor to MDCPS.

ARTICLE 3 - CONTRACTOR

3.1 **DEFINITION**

3.1.1 Contractor is the person, firm or corporation authorized to do business in the State of Florida and properly licensed or registered for the work to be performed with whom a Contract has been made with Board for the performance of the Work described in the Construction Documents. Contractor may include a Contractor acting as a Construction Manager, Design Builder, Total Program Manager or the like pursuant to the Contract Documents.

3.2 REVIEW OF CONSTRUCTION DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

3.2.1 Contractor shall carefully study and compare the Construction Documents with each other and with information furnished by Board pursuant to Subparagraph 2.2.1 and shall, within thirty (30) days following the date of the Award of the Contract, report to A/E and Board errors, inconsistencies or omissions therein. Contractor shall not be

entitled to an adjustment in the Contract Time or an adjustment in the Contract Sum if a change in the Work is required due to an error, inconsistency, omission or violation that Contractor failed to timely report that was reasonably discoverable.

- 3.2.2 Contractor shall take field measurements and verify field conditions and shall carefully compare such field measurements and conditions and other information known to Contractor with the Construction Documents before commencing the ordering of any material or doing any of the Work. The Contractor is responsible for all site conditions, utility locations, and coordination of the work with existing conditions. Errors, inconsistencies or omissions discovered shall be reported in writing to A/E at once.
- 3.2.3 Contractor shall perform the Work in accordance with the Construction Documents and approved submittals.

3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- 3.3.1 Contractor shall be solely responsible for and have control over construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Construction Documents give other specific instructions concerning these matters.
- 3.3.2 Contractor shall be responsible to Board for acts and omissions of Contractor's employees, Subcontractors and their agents and employees and other persons performing portions of the Work under the Contract and shall be responsible to A/E and Board for coordination and complete execution of the Work in accordance with the Construction Documents. The Contractor shall specifically ensure that those trades which require a Master or Journeyman to trainee ratio comply with that required ratio. This shall be strictly enforced and shall be noted daily in the Contractor's Daily Log.
- 3.3.3 Contractor shall not be relieved of obligations to perform the Work in accordance with the Construction Documents either by activities or duties of A/E in its administration of the Contract, or by tests, inspections or approvals required or performed by persons other than Contractor.
- 3.3.4 Contractor shall be responsible for inspection of portions of Work already performed under the Contract to determine that such portions are in proper condition to receive subsequent Work.
- 3.3.5 Contractor shall submit to A/E and Board's authorized representative each day a daily

work report for the preceding workday. The daily work report shall show the date, name and number of each workers by trade, including foreman, if any, employed on such date, excluding, however, all other supervisory employees whose compensation shall not be considered an element of cost for any purpose hereunder, the actual number of hours employed on the work, the character of the work that is being done and wages paid or to be paid to the worker(s). The daily report shall show the materials furnished and the amount paid or to be paid for those materials. In addition to rendering daily work reports to A/E and Board's authorized representative, Contractor shall, when any item of Work to be paid under time and material of a Work Authorization Order has been completed, render an itemized statement to Board showing the total amount expended for each class of labor and each kind of material on account of each item of such Work including each employee of the contractor or any subcontractor, sub-subcontractor or any other labor by name, hours worked per day devoted to the T & M work and the payroll records for each.

3.3.6 Contractor shall enforce strict discipline and good order among Contractor's employees and other persons carrying out the Contract including its Subcontractors and Sub-subcontractors. Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to the Contract. All of Contractor's employees and those of its Subcontractors and Sub-subcontractors shall be required to be readily identifiable as same by either badge or clothing. All subcontractors, subsubcontractors, and other levels of service providers shall be properly licensed or registered.

3.4 LABOR, MATERIALS, PRODUCTS AND EQUIPMENT

- 3.4.1 Unless otherwise provided in the Construction Documents, Contractor shall provide and pay for labor, materials, products, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, storage, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work as part of the contract price.
- 3.4.2 All labor, materials, products and equipment described in the Construction Documents shall be executed in a thoroughly substantial and worker like manner. All materials, fixtures and apparatus shall be new and first class in every respect and shall be delivered to the Site and installed and maintained in a worker like and undamaged condition. Contractor shall be responsible for the care and protection of all materials, equipment and supplies delivered to the site and all materials sold by manufacturer in sealed packages shall be so delivered to the Site. Contractor shall take particular care to ensure the work to be free from water intrusion and/or

excessive moisture.

3.4.3 In accordance with 255.20 (3) Florida Statutes, in the construction of this Project, the Contractor shall use lumber, timber, and other forest products produced and manufactured in this state whenever such products are available and their price, fitness, and quality are equal.

3.5 **SUBSTITUTIONS**

- 3.5.1 Contractor may be required by A/E or Board to furnish, in writing, a complete statement of the origin, composition and manufacturer of any materials, products or equipment to be used in the Work.
- 3.5.2 If a certain brand of materials, products or equipment is specified in the Construction Documents, it is required for the purpose of establishing a level of quality desired or purpose designated. Should Contractor find it necessary or desirable to use a material, equipment, product or system other than that specified, Contractor shall secure from Board, through A/E, written approval for the use of the alternate materials, equipment, product or system. Contractor shall make such request, in writing, only using the Board's specific separate form for Substitution Requests, not later than forty-five (45) days after the Award of the Contract and before ordering any materials requiring approval. Board is not obligated to consider Requests for Substitution or resubmittal of previously rejected substitutions after forty-five (45) days. Board is not obligated to approve Requests for Substitutions and has the discretion to require Contractor to provide the materials as specified in the Construction Documents. In no case shall Contractor be entitled to additional time and/or money arising out of Board's failure to approve Requests for Substitutions. Requests shall be submitted as follows:
 - .1 Submit five (5) copies of the Request to A/E.
 - .2 Describe in detail (complete with test reports, catalogs, brochures and black or blue line prints of drawings) the material, equipment, product or system and changes or adjustments to other Work affected. Submit samples when requested. Contractor is responsible for denoting all instances wherein the proposed substitution differs from the item specified.
 - .3 Include "cost breakdown" of item specified and of proposed substitute for which request is made. Include costs of adjustments to other work affected. Include any variation in operating, maintenance or replacement costs, and length of time product has been available on the domestic market.

- .4 State amount deducted or added to Contract amount or state "no change" in Contract amount.
- .5 State change in Contract Time for completion or state "no change" in Contract Time.
- 3.5.3 Board's decision on approval or rejection of a Request for Substitution will be final. Approval or rejection of a request to substitute will be based in part on A/E's opinion as to adaptability, durability, quality, aesthetics, contract amount change, life cycle functions or other considerations Board determines appropriate as compared to the specified or noted item(s).
- 3.5.4 Should Board, during the course of the Work, find it necessary or desirable to use a material, equipment, product or system other than specified, Board shall notify Contractor, in writing. Should Contractor accept the proposal without a change in price or time, it shall be considered an approved substitute.
- 3.5.5 If A/E requests a sample of a specified material, product, element of work or approved substitute, Contractor shall provide same. If the sample is approved, all subsequent materials used in the Work shall be equal in every respect to the sample. If the sample is not approved, Contractor shall provide an acceptable sample.

3.6 **WARRANTY**

3.6.1 Contractor warrants to Board and A/E that all products, materials and equipment furnished under the Contract will be of good quality and new unless otherwise required or permitted by the Construction Documents, that the Work will be free from defects not inherent in the quality required or permitted, and that the Work will conform with the requirements of the Construction Documents. Work not conforming to these requirements, including substitutions not properly approved and authorized, shall be considered defective. Contractor's warranty excludes remedy for damage or defect caused by abuse, modifications not executed by Contractor, improper or insufficient maintenance, by others or improper operation by others. Notwithstanding other provisions in the Contract and bid documents all warranties for all the Work shall commence when Substantial Completion is achieved for the entire Project. If required by A/E and/or Board, Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

3.7 **TAXES**

3.7.1 Contractor shall pay sales taxes for the Work or portions thereof provided by

General Conditions of the Contract for Construction - Rev. 12-5-2011

Contractor which are legally enacted when bids are received or negotiations concluded, whether or not effective or merely scheduled to go into effect.

- 3.8 PERMITS, FEES, NOTICES, JESSICA LUNSFORD ACT AND K-20 SCHOOL CODE
- 3.8.1 Contractor shall secure and pay for all **required** permits and licenses necessary for proper execution and completion of the Work in accordance with applicable law.
- 3.8.2 Contractor shall comply with and give notices required by all laws, ordinances, rules, regulations and lawful orders of public authorities bearing on performance of the Work, including but not limited to Chapter 6A-2.0010 of the Florida Administrative Code, SREF (State Requirement for Educational Facilities), Florida Building Code, OSHA, DERM and HRS and including but not limited to encounters with substances requiring special handling or specific licensure such as but not limited to asbestos, lead, mold and soil contamination.
- 3.8.3 If Contractor observes that portions of the Construction Documents are at variance with applicable laws. Contractor shall promptly notify A/E and Board, in writing, and necessary changes shall be accomplished by an appropriate Modification.
- 3.8.4 If Contractor performs any portion of the Work and knows or should have known it to be contrary to laws, statutes, ordinances, building codes, and rules and regulations without such notice to A/E and Board, Contractor shall assume full responsibility for such Work and shall bear the attributable costs.

3.8.5 **JESSICA LUNSFORD ACT (JLA)**

.1 BACKGROUND SCREENING REQUIREMENTS

In accordance with the requirements of Sections, 1012.465, and 1012.32 and 1012.467, Florida Statutes (2007), School Board Rules 6Gx13-3F-1.024 and 6Gx13-4C-1021 as amended from time to time Contractor agrees that, if Contractor receives remuneration for services, Contractor and all of its employees who provide or may provide services under this Agreement will complete criminal history checks, and all background screening requirements, including level 2 screening requirements as outlined in the above-referenced statutes and School Board rules prior to providing services to the School Board of Miami-Dade County.

Additionally, Contractor agrees that each of its employees, representatives, agents. subcontractors or suppliers who are permitted access on school grounds when students are present, who have direct contact with students or who have access to or control of school funds must meet level 2 screening requirements as described in the above-referenced statutes and School Board rules.

Pursuant to the 2007 amendments to the JLA enacted by the Florida Legislature, requirements for certain fingerprinting and criminal history checks shall be inapplicable to non-instructional contracted personnel who qualify for exemption from level 2 screening requirements as provided under Section 1012.468, Florida Statutes (2007). In addition, the provisions of Section 1012.467, Florida Statutes (2007) are incorporated herein by reference, and any provisions of this Addendum that may be inconsistent with, contrary to, or determined to be in conflict with Section 1012.467, will be superseded by said statute.

A non-instructional contractor who is exempt from the screening requirements set forth in Section 1012.465, Section 1012.468 or Section 1012.467, Florida Statutes (2007), is subject to a search of his or her name or other identifying information against the registration information regarding sexual predators and sexual offenders maintained by the Department of Law Enforcement under Section 943.043 and the national sex offender public registry maintained by the United States Department of Justice. Contractor will not be charged for this search.

Further, upon obtaining clearance by Board, if Board deems necessary, Board will issue a photo identification badge which shall be worn by the individual at all times while on Board property when students are present.

Contractor agrees to bear any and all costs associated with acquiring the required background screening - including any costs associated with fingerprinting and obtaining the required photo identification badge. Contractor agrees to require all its affected employees to sign a statement, as a condition of employment with Contractor in relation to performance under this Agreement, agreeing that the employee will abide by the heretofore described background screening requirements, and also agreeing that the employee will notify the Contractor/Employer of any arrest(s) or conviction(s) of any offense enumerated in School Board Rules 6Gx13- 3F-1.024 and 6Gx13- 4C-1.021 within 48 hours of its occurrence. Contractor agrees to provide the Board with a list of all of its employees who have completed background screening as

required by the above-referenced statutes and who meet the statutory requirements contained therein. Contractor agrees that it has an ongoing duty to maintain and update these lists as new employees are hired and in the event that any previously screened employee fails to meet the statutory standards. Contractor further agrees to notify the Board immediately upon becoming aware that one of its employees who was previously certified as completing the background check and meeting the statutory standards is subsequently arrested or convicted of any disqualifying offense. Failure by Contractor to notify the Board of such arrest or conviction within 48 hours of being put on notice and within 5 business days of the occurrence of qualifying arrest or conviction, shall constitute grounds for immediate termination of this Agreement by the Board.

The parties further agree that failure by Contractor to perform any of the duties described in this section shall constitute a material breach of the Agreement entitling the Board to terminate this Agreement immediately with no further responsibility to make payment or perform any other duties under this Agreement.

- .2 When on a school site, all Contractor's principals, employees and workers, all Subcontractor's principals, employees and workers and all other personnel engaged by the Contractor in association with the performance of the Work shall provide their Jessica Lunsford Level II Clearance Letter to the Board upon request.
- 3.8.6 Contractor agrees to comply with all sections of the Florida K-20 Education Code, Title XLVIII, Florida Statutes as it presently exists, and further as it may be amended from time to time. Further Contractor agrees that failure to comply with the Florida K-20 Education Code shall constitute a material breach of this Agreement and may result in the termination of this Agreement by the Board.

3.9 **SUPERINTENDENT**

3.9.1 Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall not be replaced without Board's prior consent. The superintendent shall represent Contractor and communications given to the superintendent shall be as binding as if given to Contractor. Important communications shall be confirmed in writing. Other communications shall be similarly confirmed on written request in each case. The superintendent and the major Subcontractors' superintendents shall be fluent in English.

3.9.2 Contractor shall at Board's request designate a replacement superintendent or other representative if Board determines that the Contractor's designated superintendent is not performing the work to Board's satisfaction.

3.10 CONTRACTOR'S PROJECT SCHEDULE

3.10.1 General

Contractor shall prepare and maintain a detailed progress schedule demonstrating fulfillment of all Contract requirements ("Schedule"), and shall utilize the Schedule for scheduling, coordinating and monitoring all Work under the Contract. Precedence format critical path method ("CPM") techniques for scheduling shall be utilized. The principles and definitions of the terms shall be as set forth in the Associated General Contractors of America, Inc. Manual, "The Use of CPM in Construction, A Manual for General Contractors and The Construction Industry," Copyright 1976; however, in the case of a discrepancy the provisions of this Article shall govern.

Copies of all schedules, including but not limited to preliminary schedule, baseline schedule and schedule updates, and as-built schedules shall be sent to MDCPS by email and on CDROM, formatted and readable by MDCPS Primavera Project Planner v 3.1 scheduling software or such other version as MDCPS may be using at the time of the Project. SureTrak and Microsoft Project software are not allowed for this project.

Details and other requirements relating to the production of schedule reports may be found in the technical specifications. If there is any conflict between the General Conditions and the technical specifications, the most stringent requirement shall apply.

Unless specifically stated otherwise in the Contract Documents or approved in writing by Board in advance, the Baseline schedule and all schedule updates shall be based on a five (5) day work week with normal daily working hours, less holidays. Contractor shall reimburse Board for all additional costs incurred by Board, including but not limited to custodial overtime for access and building code inspections, to facilitate any weekend or after hours work requested to be performed by the Contractor and approved in advance by Board.

3.10.2 Monthly Updates

Monthly updates shall be included in the monthly Requisition for Payment. If updates are not incorporated with the monthly Requisition for Payment, the monthly

General Conditions of the Contract for Construction - Rev. 12-5-2011

Requisitions will be considered incomplete and will be processed as set forth herein under Article 4.2.5. Contractor shall be responsible to prepare, submit and maintain the schedules indicated above. Contractor may be required by Board to submit a Narrative Report with each monthly update which shall include a description of current and anticipated problem areas, delaying factors and their impact, and an explanation of corrective action taken or proposed. Failure to do so may be considered a material breach of the Contract. Any additional or unanticipated costs or expense required to maintain the Schedules shall be solely the Contractor's obligation and shall not be charged to Board.

3.10.3 Adjustment of Contract Completion

The Contract Time will only be adjusted for cause specified in the Contract and/or when shown to be a delay of the critical path, or in excess of the available float in the time of the change issuance.

Contractor shall incorporate appropriate CPM activities into the schedule as change orders are approved by the Board.

All changes and/or additions to the Schedule must meet the approval of the A/E and Board.

Contractor agrees that whenever it becomes apparent from the current monthly status review meeting or the monthly computer-produced calendar-dated schedule that phasing or Contract completion dates will not be met, it will take some or all of the following actions at no additional cost to Board:

- A. Increase construction manpower in such quantities and crafts as will eliminate the backlog of Work.
- B. Increase the number of working hours per shift, shifts per working day, working days per week, the amount of construction equipment, or any combination of the foregoing to eliminate the backlog of Work.
- C. Reschedule the Work under this contract in conformance with all other specification requirements.

Prior to proceeding with any of the above actions, Contractor shall notify and obtain approval from the Board and A/E for the proposed Schedule changes. If such actions are approved, the CPM revisions shall be incorporated by Contractor into the Schedule before the next update at no additional cost to Board.

3.11 **DOCUMENTS AND SAMPLES AT THE SITE**

3.11.1 Contractor shall maintain at the Site one record copy of the Drawings, Specifications, Change Orders and other Modifications, in good order and marked currently to record changes and selections made during construction and, in addition, approved Shop Drawings, Project Schedules, Product Data, Samples and other required submittals. These shall be available to A/E and Board and shall be delivered to A/E for submittal to Board upon completion of the Work.

3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

- 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specifically prepared for the Work by Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work. Contractor or Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor shall not use the A/E's contract drawings as shop drawings, but contractor shall require these entities to produce and submit such documents independently.
- 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by Contractor to illustrate materials or equipment for some portion of the Work.
- 3.12.3 Samples are physical examples which illustrate materials, products, equipment or workmanship and establish standards by which the Work will be judged.
- 3.12.4 Shop Drawings, Product Data, Samples and similar submittals are not Construction Documents. The purpose of their submittal is to demonstrate, for those portions of the Work for which submittals are required, the way Contractor proposes to conform to the information given and the design concept expressed in the Construction Documents. Review by A/E is subject to the limitations of Subparagraph 4.2.7.
- 3.12.5 Contractor shall submit for A/E approval (copy to Owner for information) a proposed submittal schedule. This submittal schedule shall identify all required and planned submittals showing at a minimum the clear and unambiguous identity of the submittal including all shop drawings, product data samples, and similar submittals, the date they are to be transmitted to the A/E, and the date by which approval is required to maintain and coordinate with the construction schedule. The submittal schedule shall allow a minimum of 14 calendar days for review and approval/disapproval by the A/E in all cases and, where the amount or complexity of the submittal may require

additional time, it shall be noted by the A/E and accommodated by the Contractor. Submittals shall be scheduled and produced with sufficient time for appropriate review and re-submittal and re-review as required that the process shall not cause delay to the Work or any activities of the A/E or Board or separate Contractors. Contractor shall review, approve and submit to A/E all Shop Drawings, Product Data, Samples and similar submittals required by the Construction Documents with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of A/E or Board or of separate Contractors. Submittals made by the Contractor which are not required by the Construction Documents may be returned without action. Submittals shall be properly identified as specified or as A/E may require. At the time of submission, Contractor shall inform A/E, in writing, of any deviation in the submittals from the requirements of the Construction Documents. Contractor shall maintain a Shop Drawing Submittal log, copies of which shall be submitted to the A/E and Board on a weekly basis.

- 3.12.6 By approving and submitting Shop Drawings, Product Data, Samples and similar submittals, Contractor represents that it has determined and verified materials, field measurements, field construction criteria, catalog numbers and similar data, and that Contractor has checked and coordinated information contained within such submittals with the requirements of the Work and the Construction Documents. Contractor shall not be relieved of responsibility for deviations from requirements of the Construction Documents by A/E's approval of Shop Drawings, Product Data, Samples or similar submittals unless Contractor has specifically informed A/E and Board, in writing, of such deviation at the time of submittal and A/E and Board have given written approval of the specific deviation. Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by A/E's approval thereof.
- 3.12.7 Contractor shall make any corrections required by A/E and shall resubmit the required Shop Drawings, Product Data, Samples or similar submittals until approved. Contractor shall direct specific attention, in writing, or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by A/E on previous submittals.
- 3.12.8 Informational submittals upon which A/E is not expected to take responsive action may be so identified in the Construction Documents.
- 3.12.9 When professional certification of performance criteria of materials, equipment, product or system is required by the Construction Documents, A/E shall be entitled to rely upon the accuracy and completeness of such calculations and certifications.

- 3.12.10 No portion of the Work requiring a Shop Drawing, Product Data, Sample or similar submittal shall be commenced until the submittal has been approved by A/E. All such portions of the Work shall be in accordance with the approved Shop Drawings, Product Data, Samples or similar submittals.
- 3.12.11 Shop Drawings, Product Data, Samples and similar submittals shall be marked so that items which constitute information not pertaining to the equipment specified shall be so indicated. All components which are provided as "optional" by manufacturer and required hereinafter shall also be marked. Failure to comply with the preceding will result in A/E's disapproval of Shop Drawings, Product Data, Samples and similar submittals.
- 3.12.12 Shop Drawings, Product Data and Samples

The Contractor shall copy Document Control on all correspondence (incoming/outgoing) and provide copies of all plans, documents, drawings, and shop drawings for the project.

3.13 USE OF SITE

- 3.13.1 Contractor shall confine operations at Site to areas permitted by law, ordinances, and the Construction Documents and shall not unreasonably encumber the Site with materials or equipment.
- 3.13.2 Contractor shall supply additional storage on Site if required by Board for storage of furniture, appliances, equipment, materials and supplies.

3.14 **CUTTING AND PATCHING**

- 3.14.1 Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly.
- 3.14.2 Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of Board or separate contractors by cutting, patching or otherwise altering such construction, or by excavation. Contractor shall not cut or otherwise alter such construction by Board or a separate contractor except with written consent of Board and of such separate contractor; such consent shall not be unreasonably withheld. Contractor shall not unreasonably withhold from Board or a separate contractor Contractor's consent to cutting or otherwise altering the Work.

3.15 **CLEANING UP AND SALVAGE**

- 3.15.1 Contractor shall keep the premises and surrounding areas free from accumulation of waste material or rubbish caused by operations under the Contract and shall maintain the premises in a clean, safe manner. Except as otherwise provided in the Construction Documents, the Contractor shall leave the structure(s) and Site clean upon completion of the Work. The roofs, floors, gutters, glass and fixtures shall be free from soil, cement and/or paint. Nothing will be accepted that has been dented, chipped or broken during the construction, or that may have been stained, scratched or discolored. This final cleanup must be performed within the Contract Time.
- 3.15.2 If Contractor fails to clean up as provided in the Construction Documents, Board may do so and the cost thereof may be deducted from payment due or to become due.
- 3.15.3 All removed salvage items and materials (except any such stipulated as "not wanted" by Board) shall be delivered by Contractor to Board's maintenance warehouse location to be designated upon Contractor's advance notification. Telephone notification at (305) 995-4010 shall be given at least twenty four (24) hours prior to delivery.
- 3.15.4 Contractor shall prepare itemized receipts, in quadruplicate or more if required, giving pertinent description(s) of items and materials delivered. Receipts shall be signed by the Maintenance Receiving Department which shall retain one receipt. Contractor shall retain one receipt and shall distribute the other receipts to applicable Subcontractor(s) and A/E.
- 3.15.5 Removed salvage items and materials which Board advises Contractor are not wanted are the property of Contractor and shall be removed from the Site within a reasonable time or within twenty four (24) hours after written notification by the Board that the items and/or materials are interfering with operations at the Site.

3.16 **ACCESS TO WORK**

3.16.1 Ingress or egress to any portion of the Site for the delivery of any and all materials, equipment and supplies by Contractor, Subcontractor, Sub-subcontractor, manufacturer, supplier, distributor, laborer, or any other person, shall be permitted in strict compliance with the directives, rules, conditions and requirements of Board, as directed by A/E and coordinated through the Board's representative.

3.17 **ROYALTIES AND PATENTS**

3.17.1 Contractor shall pay all royalties and license fees. Contractor shall defend suits or claims for infringement of patent rights and shall hold Board harmless for loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the Construction Documents. However, if Contractor knows or has reason to believe that the required design process or product is an infringement of a patent, Contractor shall be responsible for such loss unless such information is timely furnished to A/E in writing prior to utilization.

3.18 **INDEMNIFICATION**

- 3.18.1 A. The Contractor shall indemnify and hold harmless the Owner, its officers, employees and agents, from liabilities, damages, losses, and costs, including, but not limited to, reasonable attorneys' fees, expert fees and other consultant fees, to the extent caused by the negligence, recklessness, or intentionally wrongful conduct of the Contractor and persons employed or utilized by the Contractor in the performance of this Contract.
 - B. The remedy provided to an indemnitee by Paragraph A shall be in addition to and not in lieu of any other remedy available under this Contract or otherwise.
 - C. The remedy provided to an indemnitee by this Contract shall survive this Contract and shall not be limited in any manner by acceptance, final completion, or final payment.
 - D. A claim for indemnify pursuant to this Contract shall be commenced within the period established under Florida law for commencement of an action founded on the design, planning, or construction of an improvement to real property.

The provisions of this Article are severable and if, for any reason, any one or more of the provisions contained in the Article shall be held by a court of competent jurisdiction to be invalid, illegal, against public policy, or unenforceable in any respect, the invalidity, illegality, being against public policy, or unenforceability shall not affect any other provision of this Article which shall remain in effect and be construed as if the invalid, illegal, against public policy, or unenforceable provision had never been contained in the Article.

- 3.18.2 In any and all claims against an indemnitee by any employee of Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, the indemnification pursuant to Paragraph 3.18 shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for Contractor or any Subcontractor under workers' compensation acts, disability benefit acts or other employee benefit acts.
- 3.18.3 The remedy provided to an indemnitee by this indemnification shall be in addition to and not in lieu of any other remedy available under the Contract or otherwise. Except as specifically provided in Clause 3.18.2.2, this indemnification obligation shall not be diminished or limited in any way to the total limits of insurance required in the Contract or otherwise available to Contractor or any Subcontractor. The limit on the amount of the indemnity pursuant to Subparagraph 3.18.2 applies only to the contractual indemnity resulting from Paragraph 3.18. The limit on the amount of the indemnity pursuant to Subparagraph 3.18.2 is not applicable to any right to indemnity or contribution or any other remedy which would exist in favor of an indemnitee in the absence of Paragraph 3.18.
- 3.18.4 The remedy provided to an indemnitee shall survive the Contract and shall not be limited in any manner by Acceptance, Final Completion or final payment. A claim for indemnity pursuant to Paragraph 3.18 shall be commenced within the period established under Florida law for commencement of an action founded on the design, planning, or construction of an improvement to real property.
- 3.18.5 Contractor hereby acknowledges the receipt of Ten Dollars (\$10.00) and other good and valuable consideration as part of the Contract Sum from Board and A/E which has been paid to Contractor as specific consideration for the indemnification provided herein and is incorporated as a deduction in the bid amount.

3.19 **TESTING + INSPECTION**

- 3.19.1 The Board may, at its sole discretion, identify additional tests which shall be performed by the Contractor on the project or parts of the project as directed. Such testing may include but is not limited to water intrusion, building envelope security, various roofing, and building moisture content testing.
- 3.19.2 Where such testing is required, it shall be the responsibility of the Board to procure and pay for such testing services from properly certified testing firms contracted with the Board. It shall further be the responsibility of the Contractor to schedule and ensure the proper conduct of such testing at times when that testing is appropriate in

the construction of the project.

- 3.19.3 The Contractor shall include such testing in the Contractor's construction schedule when first submitted to the Board if such tests are designated at the time of award, or, if such testing is determined to be necessary after award, the Contractor shall include same in the construction schedule at its next regular monthly submittal of that schedule with its next regular Application for Payment after receiving direction to include such testing.
- 3.19.4 The Contractor shall coordinate with the Architect and testing firm to ensure the appropriateness of the schedule for each required test and the availability of the Architect's appropriate representatives to observe such tests. The Architect shall observe, certify the proper testing procedures, and certify the results as either acceptable or unacceptable.
- In the event that the project or any portion thereof does not satisfactorily perform when any required test is administered, the Contractor shall immediately take such action as is required to correct the condition(s) causing the unacceptable performance and re-test at the Contractor's own expense until the project or that portion of the project performs acceptably when tested. Payment to the Contractor shall be adjusted appropriately in the event of any unacceptable test results.

3.20 ENERGY EFFICIENT AND SUSTAINABLE BUILDINGS

- 3.20.1 To the extent set forth herein below, the Contractor shall comply with certain of the provisions set forth in U.S. Green Building Counsel's Leadership in Energy and Environmental Design (LEED) Reference Guide for New Construction, Version 2.2 as follows:
- 3.20.2 The Contractor shall develop and implement an Erosion and Sedimentation Control (ESC) Plan for all construction activities associated with the Project. The ESC Plan shall conform with and follow the requirements of Sustainable Sites, Prerequisite 1, Construction Activity Pollution Prevention.
- 3.20.3 The Contractor shall develop and implement an Indoor Air Quality Management Plan for the construction and pre-occupancy phases of the Project that conforms with and follow the requirements of Indoor Environmental Quality Credit 3.1 Construction IAQ Management Plan During Construction.
- 3.20.4 The Contractor shall provide materials and products for the Project that conform with

and follow the requirements for Indoor Environmental Quality, Low Emitting Materials Credit 4.1 Adhesives and Sealants, 4.2 Paints and Coatings, 4.3 Carpet Systems and 4.4 Composite Wood & Agrifiber, to the extent that such materials and products otherwise meet all of the requirements of the Project Specifications and Contract Documents.

3.21 CONSTRUCTION SITE COMPLIANCE MONITORING

The additional procedures and contract provisions set forth herein are generally limited to monitoring that Contractors hire legally authorized workers and that Contractors pay its workers no less than the minimum wages prescribed by law, as follows:

3.21.1 Verification of Contractor's and Subcontractor's Employees and Workers as Eligible for Employment

- .1 The Contractor, at its own expense, shall ensure that all workers working on the Project Site, including all of Contractor's own workers, as well as all of its lower tier contractor's workers, are verified as eligible and legally authorized for employment pursuant to applicable Federal and State laws and regulations.
- .2 The Contractor shall perform or cause to be performed the verification of eligibility for employment of all workers on the Project site by electronically comparing employee information taken from the Form I-9 (the paper based employment eligibility verification form used for all new hires) against the records contained in an internet based database system such as the E-Verify system operated by and available through the U.S. Department of Homeland Security and Social Security Administration.
- .3 Prior to beginning work on the Project Site, the Contractor shall provide to the Board an initial list of all workers that have been verified as legally eligible for employment as set forth in subparagraph 3.21.1.2 that the Contractor is intending to utilize. The list shall include the full names of all workers in alphabetical order and be separately grouped by employer (i.e. contractor, subcontractors, etc.). The list shall also include the verification of eligibility for employment confirmation number for each employee or worker and the date the verification for each worker was issued by and through the employment verification system.

- .4 The Contractor shall maintain in its local home office and on the Project Site, the list of workers indicated in subparagraph 3.21.1.2, along with copies of the eligibility for employment verification documentation set forth in subparagraph 3.21.1.2.
- .5 At any stage during the construction of the Project, if the Contractor intends to employ workers on the Project Site in addition to those contained in the Contractor's initial or updated eligibility lists, the Contractor shall satisfy all requirements set forth herein, prior to those additional workers commencing work on the Project Site.
- .6 During the construction of the Project, the Board may periodically require the Contractor to perform or cause to be performed updated verifications of eligibility for employment of any or all workers on the Project Site.
- .7 The Board shall have the right to and may visit the Project Site at any time during construction of the Project and the Contractor shall be required to produce the Contractor's Required Documentation as set forth under subparagraph 3.21.1.4 for the Board's review, if requested by the Board's representative.
- .8 The Board's representative may require the Contractor to confirm that any or all workers currently working on the Project Site are, in fact, those persons indicated on the Contractor's eligibility list. The Board may also require the Contractor to jointly conduct walk-through inspections of the Project Site with the Board's representative to confirm that any or all workers on the Project Site are, in fact, those persons indicated on the Contractor's eligibility list.

3.21.2 Payment by Contractor and Subcontractors Of Minimum Required Wages to all Employees and Workers

- .1 Contractor, at its own expense, shall ensure that all workers working on the Project Site, including but not limited to all of Contractor's own workers, and all its lower tier contractor's workers, are paid wages no less than those required by Federal and State laws and that they are in full compliance with all requirements of law regarding worker's compensation insurance.
- .2 The Contractor shall post at the Project Site all notices required by State and Federal laws and regulations regarding minimum wages and other applicable employment or employee rights related laws. Such notices shall be

conspicuously posted in and about the employer's place or places of employment in locations readily visible and observable by all employees and workers on the Project Site, without any restrictions of access thereto. These notices shall include, but not be limited to, Federal Employee Rights Minimum Wage publication WHD 1088, Florida State Notice to Employees Florida Minimum Wage and the State of Florida-Division of Worker's Compensation Notice of Compliance.

.3 The Board shall have the right to and may visit the Project Site at any time during construction of the Project and the Contractor shall be required to produce for the Board's review the Contractor's Required Documentation as set forth under subparagraph 3.21.2.2.

3.21.3 Provisions Applicable to Verification and Compliance

- .1 The Contractor agrees to the reasonableness of the provisions contained herein and shall not assert, and the Board shall not be obligated to consider, any claim for disruption or interference of the Work or the Project associated with the Board's good faith actions related to these provisions and shall not request or make any claim, contingency adjustment or change order for additional costs, expenses or damages whatsoever in connection therewith.
- .2 The Contractor's Project Site Superintendent, Contractor's Project Manager or Contractor's other upper management staff shall participate with the Board's representative in conjunction with all construction site compliance monitoring set forth herein.

3.21.4 Complaints, Investigations and Violations

.1 Whenever the Board or the Board's representative observes or becomes aware of apparent violations of any of these provisions, the Contractor shall be notified promptly in writing of the apparent violation. The notice to the Contractor shall provide a description of the violation of these provisions and shall include such other documentation related to the violation as may have been made available to the Board at the time the apparent violation was observed. Concurrent with issuance of the written notice to the Contractor, copies of the written notice and supporting documentation shall be forwarded to other appropriate agencies and/or enforcement authorities for further investigation and action.

ARTICLE 4 - ADMINISTRATION OF THE CONTRACT

4.1 ARCHITECT OR ENGINEER (A/E)

- 4.1.1 A/E is the person lawfully licensed to practice architecture or an entity lawfully practicing architecture or engineering identified as such in the Contract and is referred to throughout the Construction Documents as if singular in number. The term A/E means A/E or its authorized representative.
- 4.1.2 Duties, responsibilities and limitations of authority of A/E as set forth in the Construction Documents shall not be restricted, modified or extended without written consent of Board. Consent shall not be unreasonably withheld.
- 4.1.3 In case of termination of employment of A/E, Board shall appoint an A/E, whose status under the Construction Documents shall be that of the former A/E.

4.2 A/E's ADMINISTRATION OF THE CONTRACT

- 4.2.1 A/E will provide administration of the Contract as described in the Construction Documents (1) during construction, (2) until final payment is due, and (3) during the guarantee period described at Paragraph 12.2. A/E will advise and consult with Board. A/E is the agent of Board only to the extent provided in the Construction Documents unless otherwise modified by written instruments in accordance with other provisions of the Contract.
- A/E will visit the site at least weekly or more often as appropriate to the stage of construction to become familiar with the progress and quality of the completed Work and to determine, in general, if the Work is being performed in a manner indicating that the Work, when completed, will be in accordance with the Construction Documents. A/E will not be required to make exhaustive or continuous on-site inspections to check quality or quantity of the Work. On the basis of on-site observations, A/E and its engineers and other consultants will keep Board informed of the progress and quality of the Work and will endeavor to guard Board against defects and deficiencies in the Work.
- 4.2.3 A/E will not have control over or charge of and will not be responsible for construction means, methods, techniques, sequences or procedures or for safety measures and programs in connection with the Work since these are solely Contractor's responsibility as provided at Sub-paragraph 3.3.1. While A/E is not responsible for Contractor's Project Schedules or failure to carry out the Work in accordance with the

Construction Documents nor does A/E have control over or charge of acts or omissions of Contractor, Subcontractor, or their agents or employees or any other persons performing portions of the Work, A/E will enforce the faithful performance of Contract and assure that the Work has been or is being performed in accordance with the Construction Documents. Despite A/E's lack of control and/or responsibility for the aforesaid, A/E will provide written notice to Board if it observes or has reason to become aware of any defect or non-conformance with the Construction Documents.

- 4.2.4 Communications Facilitating Contract Administration. Except as otherwise provided in the Construction Documents or when direct communications have been specially authorized by Board, Board and Contractor shall endeavor to communicate through A/E. Communications by and with A/E's engineers and other consultants shall be through A/E. Communications by and with Subcontractors and material suppliers shall be through Contractor. Communications by and with separate contractors shall be through Board.
- 4.2.5 Based on A/E's Site observations and evaluation of Contractor's notarized Requisitions for Payment, the Schedule of Values, subcontractor partial releases and the Project Schedule, A/E shall review the Requisitions for Payment and certify to Board the amounts due Contractor. Prior to issuing certification for payment, A/E shall review the status of Contractor's Construction Documents and Project Schedule and verify that the documents and/or schedules are up-to-date and accurate to the extent visual observation of the construction will disclose. A/E shall also confirm that. after the first application for payment, each subsequent application shall be accompanied by subcontractor partial lien releases fully accounting for subcontractor payments due for the previous application. If the Construction Documents and Project Schedule are not up-to-date and/or accurate, A/E shall include in its certification for payment a statement that the Construction Documents and/or Project Schedule are not up-to-date. In such event, Board may (a) hold an additional 10% of amount then due Contractor until A/E verifies that the Construction Documents and/or Project Schedule are up-to-date and accurate, (b) refuse to process the Partial or Final Requisition for Payment, or (c) pay Contractor. Contractor shall provide A/E with Contractor's and Subcontractors' partial releases of claim for provision to Board prior to processing the next month's Requisition for Payment. The A/E's certification is a representation by A/E to Owner that all required items noted herein are submitted and proper and serves as a recommendation for payment only.
- 4.2.6 A/E will have authority to recommend to Board the rejection of Work which does not conform to the Construction Documents. A/E will recommend that Board stop construction pursuant to Paragraph 2.3 when, in A/E's professional judgment, defects

and deficiencies in the construction warrant that it be stopped. Whenever A/E considers it necessary or advisable for implementation of the intent of the Construction Documents, A/E will have authority to recommend to Board additional inspection or testing of the Work in accordance with Subparagraph 13.5.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of A/E nor a decision made in good faith to exercise or not to exercise such authority will give rise to a duty or responsibility of A/E to Contractor, Subcontractors, material and equipment suppliers, their agents or employees or other persons performing portions of the Work.

A/E will review, evaluate and take appropriate action upon Contractor's submittals, such as Shop Drawings, Product Data and Samples, for conformance with the information given and the design concept expressed in the Construction Documents. A/E's approval of a separate item shall not indicate approval of an assembly in which the item functions. A/E's action will be taken with such reasonable promptness as to cause no delay in Work or in the activities of Board, Contractor, or separate Contractors, and in accordance with the Contractor's approved submittal schedule while allowing sufficient time in A/E's professional judgment to permit adequate review, but in any event, A/E's action will be taken not later than 14 days from Contractor's submittal.

If A/E is required to review a submittal more than two (2) times due to incomplete or incorrect submittals by Contractor, an appropriate Change Order may be issued by A/E deducting a sum reasonably sufficient to compensate A/E from payments due or to become due Contractor as compensation for A/E's additional expenses and services made necessary by Contractor's submission of incomplete or incorrect submittals.

Review of submittals by A/E is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of Contractor as required by the Construction Documents. A/E's review of Contractor's submittals shall not relieve Contractor of the obligations under Paragraphs 3.3, 3.5 and 3.11. A/E's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by A/E, of any construction means, methods, techniques, sequences or procedures. A/E's acceptance of a specific item shall not indicate approval of an assembly of which the item is a component.

4.2.8 Change Orders will be prepared as provided at Article 7 for Board's approval and

execution in accordance with the Construction Documents. If Contractor submits an excessive number of requests for Change Orders, unless the excessive number of requests arises from A/E errors and/or omissions in the Construction Documents, the cost thereof, including compensation for A/E's additional services and expenses made necessary by such excessive number of requests for Change Orders, may be deducted from any amount due or to become due Contractor from Board.

- 4.2.9 A/E will review and evaluate substitutions proposed by Contractors for conformance with the Construction Documents and will recommend substitutions only when it can be demonstrated that the item specified or its successor is no longer available for purchase or that the substitution provides a clear benefit to Board in performance and/or price. A/E will not be required to take any action on requests for substitutions that are not submitted to it within forty-five (45) days after the execution of the Contract except for good cause shown or unless requested by Board and, if so, the cost thereof, including compensation for A/E's additional services and expenses made necessary by such request for substitutions may be deducted from any amount due or to become due Contractor from Board.
- 4.2.10 Upon written notice from Contractor that the Work is substantially complete as defined at Paragraph 9.8.1, A/E will inspect the Project as described at Paragraph 9.8.
- 4.2.11 Upon Contractor's written notice by execution of the OEF 110[B] (Board's FM5463 equivalent) form that the Work is substantially complete, A/E will inspect the Work to determine if it is ready for inspection. If A/E believes that the Work is substantially complete, A/E will provide written notice to Board and Contractor. A/E and Board then will conduct inspection(s) and take subsequent action as described at Paragraph 9.8.
- 4.2.12 Upon written notice from Contractor, A/E and Board will conduct inspection(s) to determine the date of Final Completion as described at Paragraph 9.10. If A/E is required to conduct more than two (2) Final Completion inspections because of the failure of Contractor to complete punch list items in accordance with the Construction Documents, the cost thereof, including compensation for A/E's additional services and expenses made necessary by the failure of Contractor to timely complete all punch list work may be deducted from any amount due or to become due Contractor from Board.
- 4.2.13 Contractor shall provide A/E, which will confirm completeness and correctness of same and forward to Board for Board's review and records, the written documents required by the Construction Documents to be provided such as Warranties,

Operation and Maintenance Manuals, as-built drawings, releases of claim and other documents required of Contractor. A/E will process any pending Change Order requests and evaluate the assessment of liquidated damages, if any. Upon its determination that Contractor has fulfilled the requirements of the Construction Documents. A/E will issue a final Certificate for Payment.

- 4.2.14 If Board at its option and A/E agree, A/E will provide one (1) or more project representatives to assist in carrying out A/E's responsibilities at the Site. The duties, responsibilities and limitations of authority of such Project representatives shall be set forth in an exhibit to be incorporated in the Construction Documents.
- 4.2.15 A/E will interpret and decide matters concerning performance pursuant to the Construction Documents on written request of either Board or Contractor. A/E's response to such requests will be made with reasonable promptness, and if no agreement is made concerning the time within which interpretations required of A/E shall be furnished, A/E will furnish such interpretations within fifteen (15) days after written request.
- 4.2.16 Interpretations and decisions of A/E will be consistent with the intent of and reasonably inferable from the Construction Documents and will be in writing or in the form of drawings.
- 4.2.17 A/E's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Construction Documents and acceptable to Board.

4.3 **CLAIMS AND DISPUTES**

- 4.3.1 A Claim is a demand or assertion by one of the parties seeking, as a matter of right, adjustment or interpretation of Contract terms, payment of money, extension of time or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between Board and Contractor arising out of or relating to the Contract. Claims must be made by written notice. The responsibility to substantiate claims shall rest with the party making the Claim.
- 4.3.2 Decision of A/E. Claims, including those alleging an error and/or omission by A/E, shall be referred initially to A/E for action as provided in Paragraph 4.4. A decision by A/E shall be required as a condition precedent to litigation of a Claim between Contractor and Board as to all such matters arising prior to the date final payment is due, regardless of whether such matters relate to execution and progress of the Work. The decision by A/E in response to a Claim shall not be a condition precedent

to litigation in the event (a) the position of A/E is vacant, (b) A/E has not received evidence or has failed to render a decision within agreed time limits, or (c) forty-five (45) days have passed after the Claim has been referred to A/E. In the event that Contractor disagrees with a decision of A/E, Contractor shall immediately notify Board, in writing, through A/E, setting forth the reasons and objections to such decision.

- 4.3.3 Time limits on Claims. Claims by the Contractor must be made within seven (7) days after occurrence of the event giving rise to such Claim or within seven (7) days after the claimant first recognizes the condition giving rise to the Claim, whichever is later. No Claims will be considered unless submitted in a timely manner as required by this paragraph.
- 4.3.4 Continuing Contract Performance. Pending final resolution of a Claim, unless otherwise agreed in writing, Contractor shall proceed diligently with performance of the Contract and Board shall continue to make payments in accordance with the Construction Documents.
- 4.3.5 Waiver of Claims: Final Payment. The making of final payment shall constitute a waiver of Claims by Contractor except those expressly identified in writing to Board.
- 4.3.6 Claims for Concealed or Unknown Conditions. If conditions are encountered at the site which are (1) subsurface or otherwise concealed physical conditions which differ materially from those indicated in the Construction Documents or (2) unknown physical conditions of an unusual nature, which differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Construction Documents, and, both of which are not reasonably inferable from the Contract Documents and could not have been observable or reasonably inferable by a contractor's inspection of the site, then notice by Contractor shall be given to A/E promptly before conditions are disturbed and in no event later than three (3) days after first observance of the conditions. Contractor shall document the conditions by providing A/E with photographs and written description of the conditions. A/E will promptly investigate such conditions and will promptly notify the Board and Contractor, if they differ materially and may cause an increase or decrease in Contractor's cost of, or time required for, performance of any part of the Work. If A/E determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, A/E shall so notify Board and Contractor in writing, stating the reasons. Claims by the Contractor in opposition to such determination must be made within seven (7) days after A/E has given notice of the decision. If

Board and Contractor cannot agree on an adjustment in the Contract Sum or Contract Time, the adjustment shall be referred to A/E for initial determination, subject to further proceedings pursuant to Paragraph 4.4.

- 4.3.7 Claims for Additional Costs. No Claims for changes in the Work will be allowed by A/E unless authority for same, in writing, has been obtained from Board. If Contractor wishes to make Claim for an increase in the Contract Sum, written notice, as provided herein, shall be given before proceeding to execute the Work. If Contractor or any Subcontractor should start on the Work prior to obtaining written approval, it will be construed as an acceptance by Contractor of such Work being required under the Contract and no future Claim for additional cost will be considered or allowed by Board. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Paragraph 10.3. If Contractor believes additional cost is involved for reasons including but not limited to (1) a written interpretation from A/E, (2) an order by Board to stop the Work where Contractor was not at fault, (3) a written order for a minor change in the Work issued by A/E, (4) failure of payment by Board, (5) termination of the Contract by Board, (6) Board's suspension or (7) other reasonable grounds, Claim shall be made in accordance with the procedure established herein. All Claims made for additional costs shall be documented by supporting information thoroughly describing the basis of the Claim, the amount of Claim, and including cost information and data substantiating the amount Claimed.
- 4.3.8 Claims for Additional Time
- 4.3.8.1 If Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided for at Subparagraph 8.3.2 shall be given within the time period set forth in Article 4.3.3 herein.
- 4.3.8.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time and could not have been reasonably anticipated, and that weather conditions had an adverse effect on the scheduled construction. A claim for additional time due to rain will not be considered until and unless the time lost exceeds the twenty (20) day rain allowance provided for at Subparagraph 8.3.1.7.
- 4.3.9 Injury or Damage to Person or Property. If either party to the Contract suffers injury or damage to person or property because of an act or omission of the other party, or any of the other party's employee's or agents, or of others for whose acts such party is legally liable, written notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding seven (7) days

after first observance. The notice shall provide sufficient detail to enable the other party to investigate the matter. If a Claim for additional costs or time related to this Claim is to be asserted, it shall be filed as provided in Subparagraphs 4.3.7 or 4.3.8.

- 4.3.10 Claims for Consequential Damages. The Contractor and Board waive Claims against each other for consequential damages such as, or similar to, the following, which are caused by damage to, delay of, or termination in accordance with Article 14, of the Project. This mutual waiver includes:
 - .1 damages incurred by the Board but is not limited to: for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
 - .2 damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, loss of bonding capacity, business and reputation, and for loss of profit except anticipated profit arising directly from the Work performed.

Except as indicated above, this mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 14. Nothing contained in this Subparagraph 4.3.10 shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents. Articles 4.3.10 shall not preclude the Board from asserting consequential damages claim against a Performance Bond Surety who fails to perform its obligation under the Performance Board.

4.4 RESOLUTION OF CLAIMS AND DISPUTES

- 4.4.1 All matters in dispute under this Contract and/or the Contract Documents shall be resolved in the sole and exclusive venue of the Circuit Court for the 11th Judicial Circuit, In and For Miami-Dade County, Florida. Contractor shall insure each subcontract identifies this court as the venue for all disputes and that flow-down clauses are utilized by subcontractors, sub-subcontractors, and suppliers regarding this venue. Contractor shall include jury waiver clauses in each subcontract.
- 4.4.2 The parties hereby waive any and all right to trial by jury in an action hereunder commenced by either party in respect to this Contract or any matter arising out of this Contract, or any matter in relation to work, labor, services or materials furnished to the Project.

- 4.4.3 If a dispute arises between Board and Contractor regarding the scope of the Contractor's Work, or in the interpretation of the Contract Documents, and the parties hereto cannot promptly resolve that dispute, Board may order, in writing, the disputed work to be performed by Contractor and the dispute shall be resolved as provided at Paragraph 4.4.1.
- 4.4.4 If a dispute arises between Board and Contractor regarding the Contract Documents or the breach thereof, Contractor shall continue to prosecute the Work and maintain its progress, unless requested by Board to suspend or delay the Work or any part thereof, or unless Contractor is terminated pursuant to Paragraph 14.2.

ARTICLE 5 - SUBCONTRACTORS

5.1 **DEFINITIONS**

- 5.1.1 A Subcontractor is a person or entity other than a materialman or laborer who enters into a subcontract with Contractor for the performance of any part of Contractor's Work. The term "Subcontractor" is referred to throughout the Construction Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor.
- A Sub-subcontractor is a person or entity other than a materialman or laborer who enters into a sub-subcontract with a Subcontractor for the performance of any part of such Subcontractor's contract. The term "Sub-subcontractor" is referred to throughout the Construction Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor. The term "Sub-subcontractor" does not include separate subcontractors of a separate contractor.

5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

- As stated in the Instructions to Bidders, Contractor shall furnish in its proposal to Board the list of Subcontractors, Sub-subcontractors and materialmen (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work.
- 5.2.2 If Board has a reasonable objection to any proposed Subcontractor,

Sub-subcontractor or materialman, Board will promptly notify Contractor.

- 5.2.3 Contractor shall not contract with any proposed Subcontractor, Sub-subcontractor or materialman to whom Board has made a reasonable objection pursuant to Subparagraph 5.2.2.
- 5.2.4 If Board has a reasonable objection to a Subcontractor, Sub-subcontractor or materialman, Contractor shall propose another to whom Board has no reasonable objection.
- A Subcontractor, Sub-subcontractor or materialman may be added to the list of Subcontractors, Sub-subcontractor or materialmen if required for performance of Change Order Work. The addition of the Subcontractor, Sub-subcontractor or materialman will become a part of that Change Order Work. A listed Subcontractor, Sub-subcontractor or materialman may be changed only upon written approval of Board. The request to change shall state reasons for the request and shall be accompanied by a signed and notarized release from the listed Subcontractor or Contractor's Affidavit to Board that such a release is not obtainable.

5.3 **SUBCONTRACTUAL RELATIONS**

- 5.3.1 Contractor shall require each Subcontractor, to the extent of the Work to be performed by each Subcontractor, to be bound to Contractor by the terms of the Construction Documents, and to assume toward Contractor all obligations and responsibilities which Contractor, by the Construction Documents, assumes toward Board and A/E, excluding any obligation of the Contractor to require Subcontractors (or Subcontractors to require Sub-Subcontractors) to indemnify the Board and the A/E. Where appropriate, Contractor shall require each Subcontractor to enter into similar agreements with its Sub-subcontractors. Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreements, copies of the Construction Documents to which the Subcontractor will be bound and, upon written request of the Subcontractor, identify to Subcontractor terms and conditions of the proposed subcontract agreement which may be at variance with the Construction Documents. Subcontractors shall similarly make copies of applicable such documents available to their respective portions of proposed Sub-subcontractors.
 - .1 Subcontractor's combined overhead and profit for Change Order Work shall not be in excess of that allowed by the Board for the Contractor: fifteen percent (15%) up to \$10,000 and ten percent(10%) over \$10,000.

- 5.3.2 Contractor shall fix the scope of that portion of the Work to be performed by each of the Subcontractors. Contractor shall be held liable to Board for the performance of all of the Work and the Construction Documents do not attempt to fix the scope of the Work or responsibility of any Subcontractor. No Subcontractor shall, under any conditions, relieve Contractor of liabilities and obligations under this Contract and Contractor shall be solely responsible to Board as provided for herein.
- 5.3.3 Any disputes which may arise between Contractor and any Subcontractor or between any Subcontractor and its Sub-subcontractors must be resolved between those parties. Board and A/E will not undertake or be responsible for the resolution of such disputes.
- 5.3.4 Contractor shall be liable to Board for materials furnished as provided in Article 3. Materials shall include all materials whether manufactured and/or fabricated by other persons. In the event that an agent or other representative of Board approves the installation or erection of any item of material and Contractor concludes that same is not fabricated in good workerlike manner, Contractor shall immediately advise Board in writing.
- 5.3.5 Subcontractor shall, at Board's request to Contractor, designate a replacement representative if Board determines that the subcontractor's designated representative is not performing the work to Board's satisfaction.

5.4 **CONTINGENT ASSIGNMENT OF SUBCONTRACTS**

- 5.4.1 Each subcontract agreement for a portion of the Work is assigned by Contractor to Board provided that:
 - .1 Assignment is effective only after termination of the Contract by Board for cause pursuant to Paragraph 14.2 and only for those subcontract agreements which Board accepts by notifying the Subcontractor in writing; and
 - .2 Assignment is subject to the prior rights of the surety obligated under the Bond relating to the Contract.

ARTICLE 6 - CONSTRUCTION BY BOARD OR BY SEPARATE CONTRACTORS

6.1 BOARD'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

- 6.1.1 Board reserves the right to perform construction or operations related to the Project with Board's own forces and to award separate contracts in connection with other portions of the Project or other construction or operations on the Site under these or similar Conditions of the Contract. If Contractor claims that delay or additional cost is involved because of such action by Board, Contractor shall make such Claim as provided in Article 4 of the General Conditions.
- When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "contractor" in the Construction Documents in each case shall mean the contractor who executes each separate Board-Contractor agreement.
- 6.1.3 Board shall provide for coordination of the activities of its own forces and of each separate contractor with the Work of Contractor, who shall cooperate with them. Contractor shall participate with other separate contractors and Board in reviewing their Project Schedules when directed to do so. Contractor shall make any revisions to the Project Schedule deemed necessary by Board. The Project Schedule shall then constitute the schedule to be used by Contractor, separate contractors and Board until subsequently revised.
- 6.1.4 Unless otherwise provided in the Construction Documents, when Board performs construction or operations related to the Project with its own forces, Board shall be deemed to be subject to the same obligations and to have the same rights which apply to Contractor under the Conditions of the Contract, including, without excluding others, those stated at Article 3, this Article 6 and Articles 10, 11 and 12.

6.2 **MUTUAL RESPONSIBILITY**

- 6.2.1 Contractor shall afford Board and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities and shall connect and coordinate Contractor's construction and operation with theirs as required by the Construction Documents.
- 6.2.2 If part of Contractor's Work depends for proper execution or results upon construction or operations by Board or a separate contractor, Contractor shall, prior to proceeding

with that portion of the Work, promptly report to A/E apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of Contractor to so report shall constitute an acknowledgment that Board's or separate contractors' completed or partially completed construction is fit and proper to receive Contractor's Work, except as to defects not then reasonably discoverable.

- 6.2.3 Costs caused by Contractor's delays or by improperly timed activities by Contractor or defective construction shall be borne by the Contractor.
- 6.2.4 Should Contractor wrongfully cause damage to the Work or property of Board or separate contractors, Contractor shall assume the responsibility and liability for such damage, promptly remedying same as provided in Subparagraph 10.2.5.
- 6.2.5 Should Contractor be caused damage by any other contractor to the Work by reason of such other contractor's failure to properly perform its contract with Board, no action will lie against Board and Board shall have no liabilities therefore, but Contractor may assert a Claim for damages against such other contractor as the third-party beneficiary under the contract between such other contractor and Board.
- 6.2.6 Claims and other disputes and matters in question between Contractor and a separate contractor shall be subject to the provisions of Paragraph 4.3, provided the separate contractor has reciprocal obligations.
- 6.2.7 Board and each separate contractor shall have the same responsibilities for cutting and patching as are described for Contractor at Paragraph 3.14.

6.3 **BOARD'S RIGHT TO CLEAN UP**

6.3.1 If a dispute arises among Contractor, separate contractors and Board as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish as described at Paragraph 3.15, Board may clean up and allocate the cost among those responsible as A/E determines it to be just.

ARTICLE 7 - CHANGES IN THE WORK

7.1 CHANGE ORDERS

7.1.1 A Change Order is a written order to Contractor executed by Board, issued after

Award of the Contract, authorizing a change in the Work or an adjustment in the Contract Sum or the Contract Time. The Contract Sum and the Contract Time may be changed only by Change Order. A Change Order signed by Contractor indicates its agreement therewith, including the adjustment in the Contract Sum and/or the Contract Time. All costs, direct or indirect, allowable under the Contract are to be included in the Change Order. No change in Contract Sum shall be requested or considered due to changes in the cost of labor, material or products encountered between the time of the bid and the time of procurement of the labor or material.

- 7.1.2 Board, without invalidating the Contract, may order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and the Contract Time being adjusted accordingly. All such changes in the Work shall be authorized by Change Order and shall be performed under the applicable conditions of the Construction Documents. No changes shall be made unless ordered in writing by Board. No claim for additions to the Contract Sum will be valid unless approved by written Change Order issued by Board and accepted by Contractor.
- 7.1.3 Contractor and/or A/E may request changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions. The Contractor shall identify, describe and estimate the cost of all changes for consideration by the Board and provide appropriate data, support documentation and other information for Board's evaluation of the change and of the Contractor's proposed costs for the change. Board may request such additional information in connection with Contractor's change proposals as Board reasonably deems necessary to properly evaluate the change.
- 7.1.4 Board shall determine the method to be used to determine the adjustment in the Contract Sum such as:
 - Acceptance by Board of a lump sum price documented by an estimate which includes but is not limited to (a) direct itemized list and cost of labor to which the actual cost of payroll benefits may be added, (b) direct itemized list and cost of material and equipment to which applicable sales tax may be added and, (c) allowance to Contractor for combined overhead and profit of fifteen (15%) percent up to and including \$10,000 and ten (10%) percent over \$10,000 to be added to the total of (a) and (b), and, (d) the actual bond premium cost increase without markup;
 - .2 By unit prices when available and accepted by Board; or

- .3 By cost on a time and material basis plus a fixed percentage for overhead and profit using the following method:
 - (a) Direct itemized list and actual cost of labor to which the actual cost of payroll benefits may be added.
 - (b) Direct itemized list and actual cost of material and equipment to which may be added the applicable sales tax.
 - (c) Allowance to Contractor for combined overhead and profit of fifteen (15%) percent up to and including \$10,000 and ten (10%) percent over \$10,000 to be added to the total of items (a) and (b). All Subcontracts shall include this allowance provision.
 - (d) The actual bond premium cost increase without markup.
 - (e) Overhead includes small tools and consumable equipment.
- 7.1.5 When unit prices have been bid and accepted by the Board, they apply to both new and reconstructive work.
- 7.1.6 Rental rates for contractor owned equipment should not exceed 75% of the published rate in the latest edition of the Associated Equipment Distributors' "Rental Rates and Specifications" or in an equivalent publication.
- 7.1.7 If none of the methods set forth in Clauses 7.1.4.1, 7.1.4.2 or 7.1.4.3 is agreed upon, Contractor, upon receipt of a written Work Authorization Order from Board, shall promptly proceed with the Work involved on a time and material basis. Contractor shall submit to Board's authorized representative each day a daily work report in duplicate for approval. The daily work report will show the name and number of each worker, including foreman, if any, employed on such extra work, excluding, however, all other supervisory employees whose compensation shall not be considered an element of cost for any purpose hereunder, the actual number of hours employed on such work, the character of the work that he/she is doing and wages paid to him/her or to be paid to him/her. The daily work report shall show the materials furnished, the amount paid or to be paid therefore, an allowance for combined overhead and profit of fifteen (15%) percent up to and including \$10,000 and ten (10%) percent over \$10,000 and the actual bond premium cost increase without markup. All Subcontracts shall include this allowance provision. In addition to rendering daily

work reports for the approval of Board's authorized representative, Contractor shall, when any item of work to be paid for on a time and material basis under a Work Authorization Order has been completed, render an itemized statement to Board showing the total amount expended for each class of labor and each kind of material on account of each item of such work, including payroll documentation substantiating payroll wage and salary amounts actually paid to employees who performed the Work.

- 7.1.8 The premium time portion of authorized overtime work shall be reimbursable in accordance with agreed upon wages only but shall not include any allowance for overhead and profit. Contractor may not seek reimbursement for premium time unless such premium time has been ordered, in writing, by Board.
- 7.1.9 For deletions in the Work, reductions shall be made in the Contract Sum based upon the actual amount of savings in the cost computed upon the same basis as for additional work.
- 7.1.10. In order to minimize the impact of delay in receiving and processing changes and obtaining approval, Board has authorized representatives to approve changes of up to but not exceeding \$50,000,00, which amount may be increased by the Board subsequent to Award of this Contract. An approval hereunder by an authorized representative is a binding commitment by Board to execute a Change Order authorizing and approving the changes. The authorized representatives and established amounts are as follows:
 - .1 Project Manager not to exceed \$10,000.00
 - .2 Executive Director not to exceed \$20,000.00
 - .3 Construction Officer not to exceed \$25,000.00
 - .4 Chief Facilities Officer not to exceed \$50,000.00
- 7.1.11 Should the Contractor believe any change in the work entitles the Contractor to additional money and/or time and an extension of the contract completion date, the Contractor shall, with regard to any such claim, comply with Paragraph 4.3.3., time limits for claims. The Contractor shall not make any additions or changes to the Change Order input form which purport to waive or modify the time permitted to make a claim as provided for by 4.3.3; nor shall the Board accept such addition or change. Any claim by the Contractor for additional time or money not made within seven (7) days (4.3.3); shall forever be waived by the Contractor.

7.2 **RESERVED FOR FUTURE USE**

7.3 **FIELD ORDERS**

7.3.1 A/E has the authority to order minor changes in the Work not involving an adjustment in the Contract Sum or an extension of the Contract Time and not inconsistent with the intent of the Construction Documents. Such changes shall be effected by written order signed by the A/E and Contractor and shall be binding on Contractor. Contractor shall carry out such written field order promptly.

ARTICLE 8 - TIME

8.1 **TIME**

- 8.1.1 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allocated in the Construction Documents for Final Completion of the Work.
- 8.1.2 The contract commencement date shall be the date of award by the Board. The Contractor shall have twenty (20) days from award by the Board to obtain and submit proof of bond and insurance as required by Article 11. Failure to submit acceptable bond and insurance information in accordance with Article 11 within the twenty (20) days shall be a material breach of contract and shall be cause for the Board in its sole discretion to rescind the contract and place any and all bid security into the funds of the School Board of Miami-Dade County, Florida as liquidated damages for such failure and take such other action at contract and at law as the Board may deem appropriate. Upon submittal of appropriate documentation as required by Article 11, within the twenty (20) days required herein, the Board shall timely issue the Contractor notification of the acceptance of such documents and the Contractor may proceed with the work. The date of commencement shall not be postponed nor the contract extended by reason of failure of the Contractor or of persons or of entities for whom the Contractor is responsible to act.
- 8.1.3 The date of Substantial Completion is the date certified by A/E in accordance with Paragraph 9.8.
- 8.1.4 The term "day" as used in the Construction Documents shall mean a calendar day unless otherwise specifically defined.
- 8.1.5 All time limits stated in the Contract Documents are of the essence of the Contract. The time for completion set forth in the Contract is a binding part of the Contract upon

which the Board may rely in planning the use of the facilities to be constructed and for all other purposes.

8.2 **PROGRESS AND COMPLETION**

8.2.1 By executing the Contract, Contractor confirms that the Contract Time is a reasonable period for performing the Work. Should Contractor fail to complete the Work or obtain Substantial Completion within the time specified at Article 2 of the Lump Sum Contract, in the GMP Amendment or Work order, as applicable, and provided Contractor has not previously obtained a Change Order pursuant to Article 7 for an increase in the Contract Time, a sum appropriate with the following schedule shall be deducted for each day of delay in attaining Substantial Completion.

The schedule is agreed upon by and between the Board and the Contractor because of the difficulty of fixing the Board's actual damages in such event. The Board and Contractor specifically agree that this is a reasonable estimate of the Board's damage.

.1 Schedule for liquidated damages:

Contract Price	<u>Liquidated Damages</u>
\$100,000 to \$500,000 \$500,000 to \$1,000,000 \$1,000,000 to \$2,000,000	\$ 50.00 + 0.1% of Contract Price \$150.00 + 0.07% of all over \$ 100,000 \$430.00 + 0.05% of all over \$ 500,000 \$680.00 + 0.03% of all over \$1,000,000
\$2,000,000 and over	.\$980.00 + 0.01% of all over \$2,000,000

- .2 Contractor acknowledges that it is not necessary for Board to prove monetary loss in order to assess liquidated damages.
- .3 Nothing in this Subparagraph shall be construed as limiting the right of Board to terminate the Contract as provided for at Article 14.
- 8.2.2 Should Contractor fail to meet the Project Schedule or its revisions or updates at any time during the course of construction, the A/E may order acceleration of the Work as required at no increase to the Contract Sum.
- 8.2.3 Contractor shall not, except by agreement or instructions of Board in writing, prematurely commence operations on the Site or elsewhere prior to (a) submission of the Contract Bond as provided in Article 11, (b) issuance of the Notice to Proceed

and, (c) the effective date of insurance required by Article 11 to be furnished by Contractor. The date of the commencement of the Work shall not be changed by the effective date of such insurance.

8.3 **DELAYS AND EXTENSIONS OF TIME**

- 8.3.1 If Contractor is delayed at any time in progress of the Work for the causes or reasons listed as follows, provided that the cause or reason is beyond Contractor's control and that Contractor is not a party to such cause or reason, and provided that Contractor has complied with all provisions of the Contract related to Claims for additional time, the Contract Time may be extended by Change Order as provided for at Article 7:
 - .1 Act or neglect of Board or its representatives.
 - .2 Separate contractor(s) employed by Board, including labor strikes against said contractor(s) if said strike delays Contractor. Refusal to cross picket lines is not a cause for delay subject to an extension of time.
 - .3 Changes ordered in the Work.
 - .4 Acts or regulations of government agencies occurring after Award of the Contract.
 - .5 Labor strike against Contractor or any of its subcontractors.
 - .6 Fire, tornado, cyclone, hurricane, earthquake, epidemics or similar catastrophes.
 - .7 Rain if the Project is without a roof and completely shut down, or with a roof and the entire Project is delayed by exterior work but in either event only if the stoppage of the Work exceeds twenty (20) days since Contractor shall include in its construction schedule a specific allowance of twenty (20) days impact to the critical path for rain.
 - .8 Subsurface conditions varying materially from subsurface information furnished prior to bidding.
 - .9 War, insurrection, riot or civil commotion.
- 8.3.2 Claims relating to time shall be made in accordance with the applicable provisions of

Paragraph 4.3. The Claim shall precisely and exactly state which portion(s) of the Work was delayed, the number of days that the Project was delayed, the cause(s) and reason(s) for the delay and shall include exhibits as evidence and proof of the cause(s) and reason(s), and the Claim shall further show how the delay directly affected the completion of the Work. Contractor shall, regardless of the cause or reason of delay, continue to prosecute all of the Work not directly affected by said cause or reason and, with respect to such portion(s) of the Work affected, will take all reasonable measures to minimize the effect of said cause of delay.

- 8.3.3 Only that portion of the delay actually affecting the completion of the Work will be considered as a legitimate time extension. All claims for time extension must be evidenced and by supported an analysis of the critical path the Contractor's approved Schedule as it existed at the time of the event.
- 8.3.4 Extensions of time shall be the Contractor's sole remedy for any and all delays. No payment or compensation of any kind shall be made to the Contractor for the Contractor's increased costs, whether direct or indirect, or for any damages because of hindrance in the orderly progress of the Work or delay from any cause in the progress of the Work, whether such hindrances or delays be avoidable or unavoidable, reasonable or unreasonable, anticipated or unanticipated. Contractor expressly agrees not to make, and hereby waives any claim for its increased costs or damages on account of any delay, obstruction, or hindrance for any cause whatsoever, including but not limited to the aforesaid causes and agrees that Contractor's sole right and remedy in the case of any delay shall be an extension of the time fixed for completion of the Construction Contract. Without limitation, the Board's exercise of its rights under the changes clauses, regardless of the extent or number of such changes, shall not under any circumstances be construed as compensable delays, it being acknowledged that the Construction Contract amount includes and anticipates any and all delays whatsoever from any cause, whether such delays be avoidable or unavoidable, reasonable or unreasonable, anticipated or unanticipated.
- 8.3.5 In the event it shall be determined by a Court of competent jurisdiction that the preceding provision is inapplicable or unenforceable for any reason or cause, then the Contractor shall be entitled to the sum of \$200 per day for each day it is actually delayed by the action of or neglect of the Board or A/E or by changes in the Work, or by any other cause of delay which is attributable to the Board or A/E and beyond the Contractor's control, avoidance or mitigation and without the fault or negligence of the Contractor and/or Subcontractor or supplier at any tier. This provision contemplates anticipated and actual loss caused by any delay and the difficulty in proving the loss.

ARTICLE 9 - PAYMENTS AND COMPLETION

9.1 THE CONTRACT SUM

9.1.1 The Contract Sum is stated in the Lump Sum Contract, GMP Amendment or Work Order and including authorized adjustments, is the total amount payable by Board to Contractor for performance of the Work under the Construction Documents.

9.2 **SCHEDULE OF VALUES**

9.2.1 At least thirty (30) days prior to the first Requisition for Payment, Contractor shall submit to A/E the final detailed Schedule of Values allocated to various portions of the Work, prepared in such form and supported by such data to substantiate its accuracy as A/E may require. This Schedule, unless objected to by A/E, shall be used as a basis for reviewing Contractor's Requisitions for Payment.

9.3 **REQUISITIONS FOR PAYMENT**

- 9.3.1 Contractor shall submit to A/E its notarized Requisition for Payment with all required documentation. The Requisition for Payment shall be completed in accordance with the Schedule of Values and five (5) copies shall be submitted along with a), the updated Project Schedule, and b), Partial Releases. Contractor shall, in addition, provide A/E with Affidavits from Contractor, Subcontractors and material suppliers showing that all have been paid in full excluding retainage for the prior months' Requisition for Payment. Payment will be made in accordance with Florida Statute 218.72. For the purposes of this contract and in accordance with Florida Statute 218.74. Contractor's payment request shall be considered received by the Board on the first Wednesday after it is marked into the office of the approving agent. In the event the Contractor's Requisition for Payment is rejected, the Contractor shall resubmit the rejected Requisition for Payment with the required corrections within ten (10) days of that rejection and shall clearly mark the re-submitted Requisition for Payment as a "Re-submittal of a previously rejected Requisition for Payment".
- 9.3.1.1 The Requisition for Payment may include requests for payment on account of adjustment in the Contract Sum only when the Change Order has been approved by Board.
- 9.3.1.2 The Requisition for Payment may not include requests for payments of amounts Contractor does not intend to pay to a Subcontractor or material supplier because of a dispute or other reason.

- 9.3.1.3 The Requisition for Payment shall provide that ten (10%) percent of each Requisition for Payment shall be retained by Board until fifty (50) percent of construction has been completed pursuant to the Contract Documents. Thereafter, five (5) percent shall be retained by the Board for each Requisition for Payment until Form FM-5463 is properly submitted and accepted, except that additional retainage may be withheld for failure to update or make accurate the construction documents or project schedule or for other good cause including but not limited to failure to complete the punch list.
- 9.3.2 Unless otherwise provided in the Construction Documents, Requisitions for Payment may include materials and equipment delivered and suitably stored at the Site for subsequent incorporation in the Work. If approved in advance by Board, payment may be similarly made for materials and equipment stored off-Site if stored in a bonded warehouse and agreed to by Board in writing establishing Board's title to such materials and equipment or otherwise protecting Board's interest, and shall include applicable insurance, storage and transportation to the site for such materials and equipment stored off-Site.
- 9.3.3 Contractor warrants that title to all Work covered by a Requisition for Payment shall pass to Board no later than the time of payment. Contractor further warrants that upon submittal of a Requisition for Payment, all Work for which Certificates for Payment have been previously issued and payments received from Board shall, to the best of Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of Contractor, Subcontractors, material suppliers, or other persons or entities capable of making a claim by reason of having provided labor, materials and/or equipment relating to the Work.

9.4 **CERTIFICATES FOR PAYMENT**

9.4.1 The A/E will, within seven (7) calendar days after receipt of the Contractor's Requisition for Payment as defined herein, either issue to Board a Certificate for Payment, with a copy to Contractor, for such amount as A/E determines is properly due, or notify Contractor and Board, in writing, of A/E's reasons for withholding Certificate for Payment in whole or in part as provided in Subparagraph 9.5.1. The approval or disapproval shall be in the office of the M-DCPS Project Manager not later than Wednesday following receipt of the Contractor's Requisition for Payment as defined in 9.3.1 above. In the event that the A/E is not available or unreasonably refuses to sign the Contractor's Requisition for Payment, the Chief Facilities Officer or designee thereof may sign such requisition in place of the A/E and issue to the Board a Certificate for Payment which shall be processed by the Board as though issued by the A/E. The Board, at its sole option, may pay Contractor without receipt of

all subcontractor releases upon A/E's representation that the Work has progressed satisfactorily to the extent consistent with the Contractor's request for payment.

9.4.2 The issuance of a Certificate for Payment will constitute a representation to Board, based on A/E's observations at the Site and on the data comprising the Contractor's Requisitions for Payment, that the Work has progressed to the point indicated and that, to the best of A/E's knowledge, information and belief based on A/E's professional judgment, that the quality of the construction is in accordance with the Construction Documents. The issuance of a Certificate for Payment will further constitute a representation that in the A/E's opinion Contractor is entitled to payment in the amount specified.

9.5 **DECISIONS TO WITHHOLD CERTIFICATION**

- 9.5.1 A/E may decide not to certify payment and may withhold a Requisition for Payment in whole or in part, to the extent reasonably necessary to protect Board, if in A/E's opinion the representations to Board required by Subparagraph 9.4.2 cannot be made. If A/E is unable to certify payment in the amount of the Requisition for Payment, A/E will notify Contractor and Board as provided in Subparagraphs 9.3.1 and 9.4.1 by issuing a written disapproval of the Requistion for payment. It shall then become the Contractor's responsibility to correct the disapproved Requisition for Payment and re-submit same as defined in 9.3.1. The A/E may withhold part of the Requisition for Payment and if so shall promptly and within the time frames identified above, issue a Certificate for Payment for the amount which the A/E is able to make such representations to the Board. A/E may also decide not to certify payment or, because of subsequently discovered evidence or subsequent observations, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in A/E's opinion to protect Board from loss resulting from but not limited to the following:
 - .1 defective Work not remedied.
 - .2 third party claims filed or reasonable evidence indicating probable filing of such claims.
 - .3 failure of Contractor to make payments properly to Subcontractors for labor, materials or equipment.
 - .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum.

- .5 damage to Board or another contractor.
- .6 reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay.
- .7 persistent failure to carry out the Work in accordance with the Construction Documents, the Project Schedule, or the A/E's direction.
- .8 failure to maintain and submit as required up-to-date and accurate as-built documents and schedules.
- .9 actual or anticipated liquidated damages in excess of retainage as provided for at Clause 8.2.1.1.

9.6 **PROGRESS PAYMENTS**

- 9.6.1 After A/E has issued a Certificate for Payment, Board shall review and if in agreement make payment in the manner and within the time provided in the Construction Documents and shall so notify A/E.
- 9.6.2 Contractor shall promptly pay each Subcontractor, upon receipt of payment from Board, out of the amount paid to Contractor on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled, reflecting percentages actually retained from payments to Contractor on account of such Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in similar manner. The Board will pay the Contractor in Miami-Dade County, the General Contractor will pay all subcontractors and suppliers in Miami-Dade County and that appropriate flow-down provisions will be used to insure all payments are made in Miami-Dade County.
- 9.6.3 A/E will, on request, furnish to a Subcontractor, a copy of the Contractor's Schedule of Values. Neither Board nor A/E shall have an obligation to pay or to see to the payment of money to a Subcontractor except as may otherwise be required by law.
- 9.6.4 Board may require a consent of surety prior to Board's making progress payments.
- 9.6.5 Payment to material suppliers shall be treated in a manner similar to that provided in Subparagraphs 9.6.2, 9.6.3 and 9.6.4.

9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by Board shall not constitute acceptance of Work not in accordance with the Construction Documents.

9.7 **TIMELY PAYMENT**

9.7.1 With regard to each Requisition for Payment from the Contractor which receives Certification for Payment from the A/E or appropriate agent of the Board, the Board has established that it will consider such Requisition for Payment received on the first Wednesday after it is marked into the office of the approving agent.

9.8 SUBSTANTIAL COMPLETION

- 9.8.1 Substantial Completion means that: 1) All Work and systems are in place, 2) All safety systems are completed, tested and certified, 3) The Work can be used by the Board for the intended purpose, 4) No defects or hazards exist in the Work which could cause injury, 5) The following documents have been submitted by the Contractor and accepted by the A/E, including, the Contractor's Substantial Completion punch list with all items corrected or explained, all shop drawings and operations and maintenance manuals; and FM-5463 (formerly OEF Form 110B) has been executed by the Contractor and the A/E; and 6) Only minor elements of the Work remain to be corrected or completed for Final Completion (Acceptance).
- 9.8.2 When the Contractor considers that the work is substantially complete, the Contractor shall conduct an inspection of the work and shall prepare a written punch list of all incomplete, unsatisfactory, or otherwise unacceptable items. The Contractor shall verify the completion/correction of all the items on the Contractor's punch list as a condition for requesting a Substantial Completion Inspection by the A/E as below. The Contractor shall also call for inspections by the permitting authority and shall have all elements of the work requiring inspection by that body fully inspected and approved and shall provide copies of the Contractor's completed punch list and Form 110B signed by the Building Official to the Architect and Owner with the substantial completion documents including but not limited those listed in 9.8.1 as a condition for requesting a substantial completion inspection by the A/E as defined by 9.8.3 below.
- 9.8.3 When Contractor considers that the Work is substantially complete, and has completed the conditions of 9.8.1 and 9.8.2 above, the Contractor shall, in writing, request a Substantial Completion Inspection by the A/E. Such written request shall include all required by 9.8.1 and 9.8.2 above. Upon receipt of the request for

Substantial Completion Inspection, the A/E shall confirm all conditions required by 9.8.1 and 9.8.2 above have been met and shall then and only then schedule with the Board and the Contractor the Owner's Substantial Completion Inspection. This inspection shall include at a minimum the A/E and A/E, all engineers and other consultants, the Board, the Contractor, the subcontractors and sub-subcontractors. While each element of the work and/or trade grouping of the work does not need to be inspected at the same time, the A/E shall be responsible for insuring that the requisite parties attend each element of the Owner's Substantial Completion Inspection and shall maintain the consolidated list of deficiencies and comments. At the conclusion of the Inspection the Work will be determined to be as follows:

- .1 Finally Complete. If the Board determines that the Work has achieved Final Completion, final payment shall be made in accordance with Paragraph 9.10.5.
- Substantially Complete. If the Board determines that the Work has achieved Substantial Completion, this is the date for contract purposes that liquidated damages end and A/E will prepare a Substantial Completion Punch List which will incorporate all punch lists prepared by participants in the walk-through(s). The Substantial Completion Punch List will be provided to Contractor within fifteen days of the Inspection. Contractor shall complete the Substantial Completion Punch List within thirty (30) days of its receipt and provide written notice to A/E that the Work is ready for final inspection and acceptance.
- .3 If the Substantial Completion Punch List is not completed and the project accepted within thirty (30) days of its receipt from A/E, the Surety may be notified by the Board. Whether or not the Surety is notified, the Contractor shall meet with representatives(s) of the Board and A/E to show good cause for failure to complete the punch list as required and to present the Contractor's schedule for punch list completion. The Board, at its sole option, may accept the Contractor's presentation and allow the Contractor to continue so long as it maintains the punch list completion schedule, or, the Board, may now or at any time the Contractor fails to maintain the punch list completion schedule or fail to complete the punch list within the time set forth herein, require the Contractor vacate the jobsite and all remaining work shall be accomplished by the Board, by its own forces or others, at contractors expense, or call upon the Surety to complete the remaining work. Contractor acknowledges and agrees that the cost of such completion will be at a premium and in excess of costs Contractor would have incurred to have the work completed as part of the project under the contract and no claim shall accrue to the Contractor therefore. Contractor also acknowledges and agrees

that failure to complete the punch list as prescribed herein shall be good and sufficient reason for the Board, at its sole discretion, to declare Contractor non-responsible for a period not to exceed three years and thereby not award Contractor any work during said period even if contractor shall be a low bidder. Contractor further waives any claim to due process or other hearing or proceeding and agrees the failure to complete the punch list as prescribed herein is sufficient in and of itself to sustain such action by Board.

- .4 Not Substantially Complete. If the Board determines that the Work has not achieved Substantial Completion, the A/E will notify Contractor in writing of the deficiencies within ten (10) days of the Inspection.
- 9.8.4 Upon Substantial Completion of the Work and upon application by Contractor and certification by A/E, Board may at its option, without prejudice to other remedies it may have, reduce the retainage withheld to an amount equivalent to one and one-half (1 ½) times the value of the Work to be determined to be defective and/or incomplete.

9.9 PARTIAL OCCUPANCY OR USE

9.9.1 Board may occupy and/or use any completed portion of the Work prior to Final Completion and/or final payment. Such occupancy or use is not a waiver by Board of its right to demand strict compliance with the terms and provisions of the Contract and does not constitute acceptance of work not complying with the requirements of the Construction Documents.

9.10 FINAL COMPLETION AND FINAL PAYMENT

- 9.10.1 Final completion occurs when the Board determines there are no incomplete or deficient items of the work, and that the contract documents have been fully complied with by the Contractor. Upon receipt of written notice that the Work is ready for final inspection and Acceptance or at the completion of the thirty (30) day punch list period, whichever is earliest, and upon receipt of a final Requisition for Payment, A/E, its engineers and other consultants, Board and Contractor shall participate in a walk-through to inspect the Work. At the conclusion of the inspection, the Work shall be determined by the Board to be as follows:
 - .1 Finally Complete. If it is determined that the Work has achieved Final Completion, final payment shall be made in accordance with Subparagraph 9.10.4.

- .2 Not Finally Complete. If it is determined that the Work has not achieved Final Completion, A/E shall prepare a Final Completion Punch List and Final Completion shall be achieved by the Board in accordance with subparagraph 9.8.2.3.
- 9.10.2 If after the preparation of an initial Final Inspection Punch List additional legitimate punch list items are identified, the Board may, at its option, address the additional punch list items as items under guarantee pursuant to Subparagraph 12.2.2.
- 9.10.3 After all Final Inspection Punch List items have been completed, A/E will recommend to Board Acceptance of the Project and make recommendations regarding Contractor's final pay request. Acceptance occurs on the date that Board determines the Project is complete including the proper and complete submittal of all warranties, manuals, and other close-out documents, and no work remains to be performed.
- 9.10.4 After the work is determined to be finally complete, the Contractor may apply for Final Payment by submitting its final Requisition for Payment which shall be accompanied by (1) Contractor's Affidavit that payroll, bills for materials and equipment and other indebtedness connected with the Work have been paid or otherwise satisfied, (2) consent of surety to final payment, (3) a certificate evidencing that all insurance required by the Construction Documents is to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least thirty (30) days prior written notice has been given to Board, (4) a written statement that Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Construction Documents, (5) other data required by Board establishing payment or satisfaction of obligations such as receipts, releases and waivers of liens, claims, security interest or other encumbrances arising out of the Contract, to the extent and in such form as may be designated by Board, and (6) a completed set of as-built plans approved by the A/E. Final payment shall not be issued until Board has received a properly completed Form 209 or its equivalent.
- 9.10.5 Final Payment to Contractor is payment in full including retainage. Final Payment may not be issued prior to Acceptance as described at Subparagraph 9.10.4.
- 9.10.6 Acceptance of Final Payment by Contractor, a Subcontractor, Sub-subcontractor or material supplier shall constitute a waiver of Claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Requisition of Payment. Such waivers shall be in addition to the waiver described at Subparagraph 4.3.5.

9.11 PROMPT PAYMENT ACT

9.11.1 Payments to the Contractor will be made pursuant to The Contract for Construction and the Local Government Prompt Payment Act, Chapter 218, Part VII of Florida Statutes (2005), and any subsequent amendments thereto ("Prompt Payment Act"). To the extent that the provisions of The Contract for Construction differ with the requirements of the Prompt Payment Act, the provisions of the Prompt Payment Act in effect on the date of the initial contract will apply.

<u>ARTICLE 10 - PROTECTION OF PERSONS AND PROPERTY</u>

10.1 SAFETY PRECAUTIONS AND PROGRAMS

- 10.1.1 The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract.
- In the event Contractor encounters on the site material reasonably believed to be asbestos, lead, mold or polychlorinated biphenyl (PCB), which has not been rendered harmless, the Contractor shall immediately stop Work in the area affected and report the condition to Board and A/E with follow-up in writing. The Work in the affected area shall not thereafter be resumed except by written agreement of Board and Contractor if in fact the material is asbestos or PCB and has not been rendered harmless. The Work in the affected area shall be resumed in the absence of asbestos or PCB, or when it has been rendered harmless, by written instructions from Board or in accordance with final determination by A/E or by resolution pursuant to Paragraph 4.4.

10.2 **SAFETY OF PERSONS AND PROPERTY**

10.2.1 Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to (1) employees on the Work and other persons who may be affected thereby, (2) the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors, (3) other property at the Site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction, (4) Students, staff and other Board personnel as well as anyone who could reasonably be effected and all related property. The use of animals is strictly prohibited by any contractor, subcontractor, etc., for the use of security purposes at any DCPS construction project

site, be it an occupied or unoccupied facility unless so waived by the Board.

- 10.2.2 Contractor warrants that they are aware and knowledgeable of and shall give notices and comply with applicable laws, ordinances, rules, regulations and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss.
- 10.2.3 Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities. The Contractor shall ensure safe means of ingress and egress for all occupied portions of the existing facility impacted by the construction of the Project.
- 10.2.4 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- 10.2.5 Contractor shall promptly remedy all damage and loss (other than damage or loss insured under property insurance required by the Construction Documents) to property referred to in Clauses 10.2.1.(2) and 10.2.1.(3), caused in whole or in part by Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Clauses 10.2.1.(2) and 10.2.1.(3), except damage or loss attributable to acts or omissions of Board or A/E or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of Contractor. The foregoing obligations of Contractor are in addition to Contractor's obligations under Paragraph 3.18.
- 10.2.6 Contractor shall designate a responsible member of Contractor's organization at the Site whose duty shall be the prevention of accidents. This person shall be Contractor's superintendent unless otherwise designated by Contractor in writing to Board and A/E.
- 10.2.7 Contractor shall not load or permit any part of the construction or Site to be loaded so as to endanger its safety.
- 10.2.8 Neither A/E nor its employees shall be liable or responsible for Contractor's failure to

comply with the provisions of Article 10. A/E shall however direct the performance of work necessary to ensure the prevention of damage, injury or loss in the event that Contractor fails to perform pursuant to the provisions of Subparagraph 10.2.

10.2.9 Board may, at its option, direct Contractor to replace any employee of the Contractor, Subcontractor, Sub-subcontractor and/or materialman who consistently fails or refuses to adhere to Board safety rules and regulations and/or who is disorderly and disruptive.

10.3 **EMERGENCIES**

In an emergency affecting safety of the students, faculty staff, or other Board personnel or property of the Board, Contractor shall act, at the Board's discretion or as a last resort in the Contractor's best judgment, to prevent threatened damage, injury or loss to said personnel and/or property. Additional compensation or extension of time claimed by Contractor on account of an emergency may be determined as provided in Subparagraph 4.3.7 and Article 7.

ARTICLE 11 - INSURANCE AND BONDS

11.1 **CONTRACTOR'S INSURANCE**

- 11.1.1 <u>Evidence of Insurance:</u> The Contractor shall not commence Work until the Contractor has procured the insurance required pursuant to this Article and such insurance has been approved by the Board. The Contractor shall provide evidence of such insurance in the following manner.
 - .1 As evidence of compliance with the insurance required by Paragraph 11.1, Contractor shall furnish the Board with a fully completed certificate(s) of insurance signed by an authorized representative of the insurer(s) providing the coverages.
 - .2 The evidence of insurance described in Clause 11.1.1.1 shall provide that the Board be given no less than thirty (30) days written notice prior to cancellation. The Notice of Cancellation shall be by Endorsement in the policies.
 - .3 Until such time as the insurance is no longer required to be maintained by Contractor, Contractor shall provide the Board with renewal or replacement evidence of the insurance in the manner described by Clause 11.1.1.1 no less than fifteen (15) days before the expiration or termination of the insurance for

which previous evidence of insurance has been provided.

- 11.1.2 <u>Qualification of Insurers</u> Insurers providing the insurance required of Contractor by this Contract must meet the following minimum requirements.
 - .1 Such insurers must either:
 - (a) authorized by subsisting certificates of authority issued to the companies by the Department of Insurance of the State of Florida or an eligible surplus lines Insurer under Florida Statute.

or

- (b) with respect only to the Workers' Compensation/Employers' Liability coverage, (1) authorized as a group self-Insurer pursuant to Florida Statute 440.87 or (2) authorized as a commercial self-Insurance fund pursuant to Florida Statute.
- .2 In addition, Insurers, other than Lloyds of London or those authorized by Florida Statute 440.87 or Florida Statute 624.482, shall have and maintain throughout the period for which coverage is required a Best's Rating of "B+" or better and a financial size Category of "IV" or better, according to the latest edition of Best's Key Rating Guide, published by the A.M. Best Company.
- .3 If, during this period when an Insurer is providing the Insurance required by this Contract, an Insurer shall fail to comply with the foregoing minimum requirements, as soon as the Contractor has knowledge of such failure, the Contractor shall immediately notify the Board and immediately replace the Insurance provided by the Insurer with an Insurer meeting the requirements.
- 11.1.3 <u>Description of Required Insurance</u> Without limiting any of the other obligations or liabilities of Contractor, Contractor shall, at Contractor's sole expense, procure, maintain and keep in force the amounts and types of insurance conforming to the minimum requirements set forth in this Article. Except as otherwise specified in this Contract, the insurance shall commence prior to the commencement of Work by the Contractor and shall be maintained in force, without interruption, until Final Completion.
 - .1 <u>Commercial General Liability Insurance</u>
 Except as otherwise provided, the Commercial General Liability Insurance provided by the Contractor shall conform to the requirements hereinafter set forth in this Clause.

- (a) Contractor's insurance shall cover Contractor for those sources of liability (including, but not by way of limitation, coverage for operations, Products/Completed Operations, independent contractors and liability contractually assumed) which would be covered by the latest occurrence form edition of the standard Commercial General Liability Coverage Form (ISO Form CG 00 01) as filed for use in the State of Florida by the Insurance Services Office.
- (b) The minimum limits to be maintained by Contractor (inclusive of any amounts provided by an umbrella or excess policy) shall be those stated in the All Other column in the Required Limits of Insurance form (Form INS 02/09).
- (c) Except with respect to coverage for Property Damage Liability, the Commercial General Liability coverage shall apply on a first dollar basis without application of any deductible or self-insured retention. The coverage for Property Damage Liability may be subject to a maximum deductible of \$1,000 per occurrence. Contractor shall pay on behalf of the Board or the Board's member, officer or employee any such deductible applicable to a claim against the Board or the Board's member, officer or employee for which the Board or the Board's member, officer or employee is insured as an Additional Insured by Endorsement..

.2 General Liability Protection For Board

Except as otherwise provided, the insurance provided by Contractor shall conform to the requirements hereinafter set forth in this Clause.

- (a) Contractor shall provide the Board with Owners Protective Liability insurance which shall cover the Board for all sources of liability which would be covered by the latest occurrence edition of the standard Owners and Contractors Protective Liability Coverage Form, Coverage for Operations of Designated Contractor (ISO Form CG 00 09), (hereinafter OCP Policy) as filed for use in the State of Florida by the Insurance Services Office, without the attachment of restrictive endorsements.
- (b) The Board shall be the Named Insured on the OCP Policy and, if applicable, the excess policy or policies. The policy or policies shall be endorsed to include the Board's members, officers and employees as

insureds. The policy or policies shall include Contractor, its Subcontractors and its Sub-subcontractors as the contractor designated in the declarations.

- (c) The minimum OCP Policy limits per occurrence and, if subject to an aggregate, annual aggregate to be provided by Contractor (inclusive of any amounts provided by excess policies) shall be the amounts shown in the All Other column in the Required Limits of Insurance form (DC-INS 02-09), as the minimum Each Occurrence and General Aggregate limits respectively required for the Commercial General Liability Coverage.
- (d) As an alternative to the OCP Policy, the Contractor may include the Board and the Board's members, officers and employees as "Additional Insureds by Endorsement" on the Commercial General Liability coverage required pursuant to Clause 11.1.3.1 - Commercial General Liability Insurance. If this alternative is selected, the coverage afforded such Additional Insureds Endorsements shall be no more restrictive than that which would be afforded by adding the Board and the Board's members, officers and employees as Additional Insureds using the latest Additional Insured - Owners, Lessees or Contractors (Form B) Endorsement (ISO Form CG 20 10). The Certificate of Insurance shall be clearly marked to reflect "The School Board of Miami-Dade County, Florida and its members, officers and employees as additional insured" for the use of this alternative.

.3 Automobile Liability Insurance

Except as otherwise provided, the Automobile Liability Insurance provided by the Contractor shall conform to the following requirements.

- (a) Contractor's insurance shall cover Contractor for those sources of liability which would be covered by Section II of the latest occurrence edition of the standard Business Auto Policy (ISO Form CA 00 01) including coverage for liability contractually assumed, as filed for use in the State of Florida by the Insurance Services Office.
- (b) Coverage shall be included on all owned, non-owned and hired autos used in connection with this Contract.
- (c) The minimum limits to be maintained by Contractor (inclusive of any

amounts provided by an umbrella or excess policy) shall be the amounts stated in the Required Limits of Insurance form (DC-INS 02-09).

- .4 <u>Workers' Compensation/Employers' Liability Insurance</u>
 Except as otherwise provided, the Workers' Compensation/Employers' Liability Insurance provided by the Contractor shall conform to the following requirements.
 - (a) Contractor's insurance shall cover Contractor (and to the extent its Subcontractors and its Sub-subcontractors are not otherwise insured, its Subcontractors and Sub-subcontractors) for those sources of liability which would be covered by the latest edition of the standard Workers' Compensation Policy, as filed for use in Florida by the National Council on Compensation Insurance, without restrictive endorsements. In addition to coverage for the Florida Workers' Compensation Act, where appropriate, coverage is to be included for the Federal Employers' Liability Act and any other applicable Federal or State law.
 - (b) Subject to the restrictions of coverage found in the standard Workers' Compensation Policy, there shall be no maximum limit on the amount of coverage for liability imposed by the Florida Workers' Compensation Act or any other coverage customarily insured under Part One of the standard Workers' Compensation Policy. The minimum amount of coverage for those coverages customarily insured under Part Two of the standard Workers' Compensation Policy (inclusive of any amounts provided by an umbrella or excess policy) shall be those amounts stated in the Required Limits of Insurance form (DC-INS 02-09).
- 11.1.4 Contractor's Insurance Primary The insurance provided by the Contractor pursuant to Paragraph 11.1 shall apply on a primary basis and any other insurance or self-insurance maintained by the Board or all Board's member, officer or employee shall be excess of and not contributing with the insurance provided by or on behalf of Contractor.
- 11.1.5 <u>Deductible Provisions</u> Except as otherwise specifically authorized by Subparagraph 11.1.3, the insurance maintained by the Contractor shall apply on a first dollar basis without application of a deductible or self-insured retention.
- 11.1.6 <u>Insurance Is Additional Remedy</u> Compliance with the insurance requirements of this

Contract shall not limit the liability of Contractor, its Subcontractors, its Subcontractors, its employees or its agents to the Board or others. Any remedy provided to the Board or the Board's members, officers or employees by the insurance shall be in addition to and not in lieu of any other remedy available under the Contract or otherwise.

- 11.1.7 <u>Insurance on Subcontractors and Sub-subcontractors</u> Contractor shall require its Subcontractors and its Sub-subcontractors to maintain any and all insurance required by law. Except to the extent required by law and as may be required by Clause 11.1.3.1(c), this Contract does not establish minimum insurance requirements for Subcontractors or Sub-subcontractors.
- 11.1.8 <u>No Waiver By Approval/Disapproval</u> Neither approval by Board nor failure to disapprove the insurance furnished by Contractor shall relieve Contractor of Contractor's full responsibility to provide the insurance as required by the Contract.
- 11.1.9 <u>Default of the Contract or for Failure to Maintain Insurance</u> The Contractor shall be in default of this Contract for failure to maintain insurance as required by Article 11.
- 11.1.10 Required Limits of Insurance form (DC-INS 02-09) The Required Limits of Insurance form (DC-INS 02-09) is contained in Article 17 -Forms.
- 11.1.11 Asbestos Special Provisions
 - .1 If the Contract Documents identify the Contract as an "Asbestos" contract, the minimum requirements for the Commercial General Liability and General Liability Protection for the Board shall be modified as follows:
 - (a) In addition to the scope of coverage otherwise specified in Clause 11.1.3.1 for Commercial General Liability Insurance, the Contractor's insurance shall cover the Contractor for those sources of liability which would be covered by the latest occurrence form edition of the standard Commercial General Liability Coverage Form (ISO Form CG 00 01) and the Hazardous Materials Contractors Endorsement (ISO Form CG 22 78) as filed for use in the State of Florida by the Insurance Services Office.
 - (b) In lieu of the limits required by Clause 11.1.3.1, the minimum limits to be maintained by Contractor (inclusive of any amounts provided by an umbrella or excess policy) for Commercial General Liability Insurance

- shall be those stated in the Asbestos Work column in the Required Limits of Insurance form (DC-INS 02-09).
- (c) In addition to the scope of coverage otherwise specified in Clause 11.1.3.2, the OCP Policy shall be endorsed to remove the exclusions applicable to the removal, replacement, repair, enclosure or encapsulation of any hazardous material or substance from a building or structure by or on behalf of the Contractor, (i.e., remove paragraphs (1)(a) and (1)(d)(ii) under Exclusion j. of Section I of the OCP Policy).
- (d) In lieu of the limits required by Clause 11.1.3.2, the minimum OCP Policy limits per occurrence and, if subject to an aggregate, annual aggregate to be provided by Contractor (inclusive of any amounts provided by excess policy) shall be those stated in the Asbestos Work column in the Required Limits of Insurance form (DC-INS 02-09), as the minimum Each Occurrence and General Aggregate limits respectively required for the Commercial General Liability Coverage.
- .2 If the Contract Documents do not identify this Contract as an "Asbestos" contract, and the Contractor encounters asbestos which has not been rendered harmless (by removal or otherwise), and, pursuant to Subparagraph 10.1.2 of the General Conditions of the Contract for Construction, the Board, by written agreement with Contractor, contracts with Contractor to render the asbestos harmless, the insurance required by Article 11 shall be amended as follows.
 - (a) Unless Contractor subcontracts the entire Work in connection with the rendering of the asbestos harmless, Contractor shall maintain throughout the duration of such Work, the minimum insurance required by Subparagraph 11.1.3 as modified by Clause 11.1.11.1 in the same manners as if the Contract Documents had identified the Contract as an "Asbestos" contract.
 - (b) If any Work in connection with the rendering of the asbestos harmless is subcontracted by Contractor, each such Subcontractor shall be licensed as an asbestos contractor. In addition, each such Subcontractor shall fully comply with Paragraph 11.1 of this Article including as modified by Clause 11.1.11.1 in the same manner as if the Subcontractor were the Contractor and the Contract Documents had

identified the Contract as an "Asbestos" contract.

- (c) If Contractor subcontracts the entire Work in connection with the rendering of the asbestos harmless with subcontractors licensed as asbestos contractors, and each such Subcontractor complies with the insurance requirements of the preceding paragraph, Contractor shall maintain throughout the duration of such Work, the minimum insurance required by Subparagraph 11.1.3 without the modifications specified in Clause 11.1.11.1.
- 11.1.12 Contractor may be required to provide certified copies of all insurance policies with Notice of Cancellation endorsements to the Board.

11.2 BOARD PROVIDED PROPERTY INSURANCE PROGRAM

11.2.1 <u>Board to Maintain Property Insurance Program:</u> Except as otherwise provided, the Board shall maintain a property insurance program on behalf of the Contractor and its Subcontractors as described in this Paragraph 11.2.

11.2.2 <u>Definitions Applicable to Paragraph 11.2</u>

- .1 "Board Commercial Insurance" For the purposes of this Paragraph 11.2, the term "Board Commercial Insurance" shall refer to the insurance policy or policies obtained from commercial insurance companies by the Board pursuant to the Board's obligation to maintain the Property Insurance Program under this Paragraph 11.2.
- "Covered Loss" For the purposes of this Paragraph 11.2, the term "Covered Loss" shall refer to loss which, but for the application of deductibles in, or limits of, the Board Commercial Insurance would be insured under the Board Commercial Insurance.
- .3 "The Board Property Insurance Program" For the purposes of this Paragraph 11.2, the term "the Board Property Insurance Program" shall refer to the combination of the Board Commercial Insurance and amounts otherwise insured by the Board, or assumed by the Board, pursuant to this Paragraph 11.2.
- 11.2.3 <u>Scope of Coverage</u> Except as otherwise provided under Subparagraphs 11.2.4, 11.2.5, 11.2.6, 11.2.7, 11.2.8 and 11.2.9, the Board Property Insurance Program shall

be subject to the exclusions and other limitations and conditions in the Board Commercial Insurance in effect at the time that Notice to Proceed is received by Contractor.

- 11.2.4 <u>Limitation to Jobsite</u> Notwithstanding any provision(s) in the Board Commercial Insurance to the contrary, the coverage under the Board Property Insurance Program for the Contractor and its Subcontractors will be limited to Covered Loss resulting from damage or destruction of property while such property is at the construction jobsite of the Project as described in this Contract.
- 11.2.5 <u>No Time Element Coverage</u> Notwithstanding any provision(s) in the Board Commercial Insurance to the contrary, the coverage under the Board Insurance Program for Contractor and its Subcontractors will not include coverage for any loss of income, business interruption, extra expense or any other time element type losses.
- 11.2.6 No Coverage on Contractor's Tools or Equipment Notwithstanding any provision(s) in the Board Commercial Insurance to the contrary, the coverage under the Board Property Insurance Program for the Contractor and its Subcontractors will be limited to property which has been, or is intended to be, incorporated into the Work as part of the contract price for which title has either passed to the Board, or is intended to pass to the Board.
- 11.2.7 <u>Limit on Covered Loss</u> Notwithstanding any provision(s) in the Board Commercial Insurance to the contrary, the limit of Covered Loss per occurrence under the Board Property Insurance Program for the Contractor and its Subcontractors, on a combined basis, shall be no more than the lesser of:
 - .1 the total compensation due the Contractor under this Contract; or
 - .2 the insurable value of the covered property at the construction jobsite of the the Project as described in this Contract; or
 - .3 except with respect to the policy limit and sublimits with respect to windstorm, in the Board Commercial Insurance, the amount of any applicable sublimit in the Board Commercial Insurance.

Except with respect to any applicable aggregate sublimit in the Board Commercial Insurance, the limit on Covered Loss will apply on an occurrence basis regardless of the number of persons or organizations (including the Contractor and its

Subcontractors) insured.

- 11.2.8 Responsibility for Deductible The Contractor shall be solely responsible for the first \$25,000 of Covered Loss in any one occurrence. If the deductible on the Board Commercial Insurance applicable to the Covered Loss exceeds the \$25,000 for which the Contractor is responsible, the Board shall be responsible for that portion of Covered Loss incurred by the Contractor and its Subcontractors in excess of \$25,000 up to the amount of the deductible in the Board Commercial Insurance.
- 11.2.9 Responsibility for Covered Loss in Excess of Board Commercial Insurance If the amount of the Covered Loss exceeds the amount of coverage available in the Board Commercial Insurance, the Board shall be responsible for that portion of Covered Loss incurred by the Contractor and its Subcontractors in excess of the amount available in the Board Commercial Insurance up to the limit of Covered Loss pursuant to Subparagraph 11.2.7.
- 11.2.10 Commencement of the Board Property Insurance Program The Board Property Insurance Program shall commence with respect to the Work at the later of the date Notice to Proceed is received by Contractor under this Contract or commencement of work at the construction jobsite of the Project as described in this Contract.
- 11.2.11 <u>Termination of Board Property Insurance Program</u> Coverage under the Board Property Insurance Program for the Contractor and its Subcontractors shall terminate at the earliest of:
 - .1 With respect to any completed portion of the Work, if, pursuant to Paragraph 9.9 of this Contract, the Board elects to occupy and/or use such completed portion of the Work prior to Substantial Completion, the date the Board first occupies or uses such completed portion of the Work; or
 - .2 the date of Substantial Completion as certified by the A/E in accordance with Paragraph 9.8; or
 - .3 if work by the Contractor is permanently abandoned or terminated prior to Substantial Completion, at the time such work is permanently abandoned or terminated; or
 - .4 termination of this Contract by the Board.
- 11.2.12 <u>Board's Right to Terminate, Modify or Replace</u> The Board reserves the right to

terminate in whole or in part or modify through the selection of different perils, terms, deductibles, whole or partial self-insurance or otherwise modify the Board Commercial Insurance. In the event of termination or modification (whether initiated by the Board or its insurers), the Board will assume the responsibility for that portion of any loss suffered by the Contractor or its Subcontractors which would have been covered by the Board Commercial Insurance.

- 11.2.13 Risk of Loss on Board Purchased Property The Board shall retain the risk of loss of, and shall bear the economic burden of obtaining insurance coverage for, damage or loss to tangible personal property which meets the criteria in Rule 12A-1.094(4)(b)1-4, F.A.C., to determine if the Board is the purchaser for the purposes of a tax exemption under Section 212.08(6), Fla. Stat.
- 11.2.14 Contractor Retains Risk of Loss Except with respect to the Board's responsibility for any excess deductible portion pursuant to Subparagraph 11.2.8, and the Board's responsibility for any Covered Loss which exceeds the amount of coverage available in the Board Commercial Insurance pursuant to Subparagraph 11.2.9, and the Board's retaining the risk of loss pursuant to Subparagraph 11.2.13, the Board's agreement to maintain the Board Property Insurance Program pursuant to this Paragraph 11.2 shall in no way transfer the risk of loss to the Board. The risk of loss shall remain with the Contractor until the earliest of:
 - .1 With respect to any completed portion of the Work, if, pursuant to Paragraph 9.9 of this Contract, the Board elects to occupy and/or use such completed portion of the Work prior to Substantial Completion, the date the Board first occupies or uses such completed portion of the Work; or
 - .2 the date of Substantial Completion as certified by the A/E in accordance with Paragraph 9.8; or
 - .3 if work by the Contractor is permanently abandoned or terminated prior to Substantial Completion at the time such work is permanently abandoned or terminated; or
 - .4 termination of this Contract by the Board.
- 11.2.15 <u>Board Property Insurance Program Subject to Limitations</u> The providing of the Board Property Insurance Program shall not constitute any representation by the Board with respect to the adequacy of the insurance to protect the Contractor or its Subcontractors against property insurance type losses. The Board emphasizes that

the coverages in the Board Property Insurance Program are limited in scope and do not necessarily include all insurance coverages, either desirable or normally maintained by the Contractors or Subcontractors. Except as otherwise specifically provided in the Contract, the providing of the Board Property Insurance Program shall not be construed to be a limitation on the nature or extent of the Contractor's or its Subcontractors' obligations under this Contract nor to relieve the Contractor or its Subcontractors of any such obligations.

- 11.2.16 Review of Policies by Contractor Upon their request, the Contractor and its Subcontractors may, at the offices of the Board during the regular business hours of such offices, review and obtain copies of the Board Commercial Insurance.
- Initial Notice of Claim under Board Property Insurance Program In addition to, and not in lieu of, any other notice required under this Contract, if a Contractor or its Subcontractor suffers injury or damage to property which might result in Covered Loss under the Board Property Insurance Program, written notice of such injury or damage shall be given to the Office of Risk & Benefits Management of the Board, as soon as practical, but not exceeding seven (7) days after first observance of such injury or damage. The notice shall provide sufficient detail to enable the Office or Risk & Benefits Management, or its designee, to provide an initial report to the Board's insurers and to properly investigate the matter. The written notice shall be directed to:

[Title of Person]
Office of Risk & Benefits Management
1500 Biscayne Blvd.
Miami, FL 33132

- 11.2.18 Cooperation of Contractor and Subcontractors The Contractor and all of its Subcontractors shall assist the Board and the Board's insurers and fully cooperate with them in connection with the reporting, investigation and adjusting of claims under the Board Property Insurance Program whether or not involving the respective Contractor or Subcontractor.
- 11.2.19 Waiver of Subrogation To the extent such insurance permits, and then only to the extent Board collects under the Board Commercial Insurance, Board waives any and all claims against Contractor and Subcontractors and their respective agents, servants and employees, for loss or damage to Board's property. To the extent such insurance permits and then only to the extent the Contractor collects under its property insurance coverage, Contractor waives any and all claims against Board and its agents, servants and employees for loss or damage to Contractor's property.

Contractor shall require all Subcontractors to waive, to the extent such insurance permits and then only to the extent such Subcontractor collects under its property insurance coverage, any and all claims against Board and its agents, servants and employees for loss or damage to such Subcontractor's property.

11.3 **BONDS**

- 11.3.1 Bonds Required. If the initial Contract price is in excess of \$200,000, within 20 days from the date of the award, Contractor shall provide Board with a Public Construction Payment Bond and Public Construction Performance and Guarantee Bond meeting the standards specified herein, on the forms provided by the Board with Power of Attorney Affidavit attached, each in an amount not less than the Contract price. The Bond will guarantee all materials and workership installed and performed under the contract for the maximum time allowed by law after final acceptance of the Project by the Board.
- 11.3.2 <u>Sureties' Qualifications</u>: All Bonds required under this Contract, including, but not by way of limitation, and Bid Bond, Public Construction Payment Bond or Public Construction Performance and Guarantee Bond, shall be written through a reputable and responsible surety bond agency licensed to do business in the State of Florida and with a surety which holds a certificate of authority authorizing it to write surety bonds in Florida meeting the following requirements:

A. Ratings by A.M. Best

Unless the Contract price is \$500,000 or less and the surety qualifies pursuant to Clause C. below, the surety company or corporation shall have the following minimum ratings by The A.M. Best Company:

Contract Amount: Best's Rating

From: \$ 200,000.00 A or better, No Minimum Class

To: \$5,000,000.00

From: \$ 5,000,000.01 A or better, Class IV

To: \$10,000,000.00

From: \$10,000,000.01 A or better, Class V

or more

B. Circular 570

Unless the Contract price is \$500,000 or less and the surety qualifies pursuant to Clause C. below, regardless of the size of the bond, in addition to meeting the requirements of Clause A. above, the surety shall also comply with the Circular 570 requirements as set forth in this Clause B. The surety shall maintain a current certificate of authority as an acceptable surety on Federal Bonds in accordance with the U.S. Department of Treasury Circular 570, current revision. If the amount of the bond exceeds the underwriting limitations set forth in the Circular, in order to qualify, the net retention of the surety company shall not exceed the underwriting limitation in the Circular and the excess risk must be protected by co-insurance, reinsurance, or other methods in accordance with Treasury Circular 297, revised September 1, 1978(3) CFR Section 223.10 - Section 223.111. Further the surety company shall provide the Board with evidence satisfactory to the Board, that such excess risk has been protected in an acceptable manner.

C. Contract Price of \$500,000 or Less

Notwithstanding the foregoing Clauses A. and B., in the event the Contract price does not exceed \$500,000, in accordance with Section 287.0935, Florida Statutes, bond with a surety company in compliance with the following requirements shall be acceptable:

- The surety company is licensed to do business in the State of Florida; and
- (ii) The surety company holds a certificate of authority authorizing it to write surety bond in Florida; and
- (iii) The surety company has twice the minimum surplus and capital required by the Florida Insurance Code at the time the invitation to bid is issued; and
- (iv) The surety company is otherwise in compliance with the provisions of the Florida Insurance Code; and
- (v) The surety company holds a currently valid certificate of authority issued by the United States Department of the Treasury under 31 U.S.C. ss. 9304 to 9308.

In order to qualify as an acceptable surety company under this Clause C., an Affidavit for Surety Company shall be executed by an Officer of the surety bond insurer as evidence that a surety company is in compliance with the foregoing requirements.

- 11.3.3 Additional or Replacement Bond. It is further mutually agreed between the parties hereto that if, at any time, the Board shall deem the surety or sureties upon any bond to be unsatisfactory, or if, for any reason, such bond (because of increases in the work or otherwise) ceases to be adequate, the Contractor shall, at its expense within five (5) days after receipt of notice from the Board so to do, furnish an additional or replacement bond or bonds in such form, amount, and with such surety or sureties as shall be satisfactory to the Board. There shall be no lapse in surety coverage and failure to do so shall be a material breach of this contract. In such event, no further payments to the Contractor shall be deemed to be due under this Contract until such or additional security for the faithful performance of the work shall be furnished in manner and form satisfactory to the Board.
- 11.3.4 <u>Florida Agent</u>: The surety company shall have a Florida agent whose name shall be listed in the prescribed space on the forms provided by the Board for all bonds required by the Board.
- Alternate Form of Security: In lieu of the Public Construction Performance Bond and Public Payment and Guarantee Bond, Contractor may, pursuant to Section 255.05, Florida Statutes, provide an alternate form of security in the form of cash, a money order, a certified check, a cashier's check, an irrevocable letter of credit, or security of a type listed in Part II of Chapter 625, Florida Statutes. Any such alternative form of security shall be for the same purpose and be subject to the same conditions as those applicable to the bond for which the alternative form of security is being substituted. The determination of the value of an alternative form of security shall be made by the Board.
- 11.3.6 Other Bonds. If the Work includes connections to infrastructure, including but not limited to water and/or sewer facilities within local and state dedicated right-of-ways, canal crossings and FDOT infrastructure, Contractor, at its expense and in a timely manner, will comply with all Bond requirements set forth by Miami-Dade County, State of Florida and other applicable jurisdictional agencies.
- 11.3.7 <u>Co-Sureties</u>: Subject to the following requirements, the bonds required by this Contract may be provided by more than one surety.

- A. Each surety may, by setting forth the limits of its liability in the bond as a definite and specified sum, limit its liability to such amount. If no such limit of liability is indicated, the surety shall be jointly and several liable for the full amount of the bond required.
- B. The entire amount of the bond must be provided by one or more sureties who meet the requirements of Subparagraph 11.3.2.
- C. If one or more of the co-sureties do not meet the requirements of Subparagraph 11.3.2, those co-sureties who do not meet the requirements of Subparagraph 11.3.2 must be jointly and severally liable for the entire amount of the bond required.
- D. If the liability of a co-surety is less than the entire amount of the bond required, the co-surety must, for the purposes only of allowing a joint action or actions against any of its co-sureties, bind itself jointly and severally with such other co-sureties.

ARTICLE 12 - UNCOVERING AND CORRECTION OF WORK

12.1 UNCOVERING OF WORK

- 12.1.1 If a portion of the Work is covered contrary to A/E's request or to requirements specifically expressed in the Construction Documents, it must, if required in writing by A/E, be uncovered for A/E's observation and be replaced at Contractor's expense without change in the Contract Time.
- 12.1.2 If a portion of the Work has been covered which A/E has not specifically requested to observe prior to its being covered, A/E may request to see such Work and it shall be uncovered by Contractor. If such Work is in accordance with the Construction Documents, costs of uncovering and replacement shall, by appropriate Change Order, be charged to Board. If such Work is not in accordance with the Construction Documents, Contractor shall pay such costs unless the condition was caused by Board or a separate contractor in which event Board shall be responsible for payment of such costs.

12.2 CORRECTION OF WORK, GUARANTEE AND WARRANTY

12.2.1 Contractor shall promptly correct Work rejected by A/E or failing to conform to the requirements of the Construction Documents, whether observed before or after Substantial Completion and whether or not fabricated, installed or completed. Contractor shall bear costs of correcting such rejected Work, including additional testing and inspections and compensation for A/E's services and expenses made necessary thereby.

- 12.2.2 Except as otherwise provided in Construction Documents, Contractor shall guarantee the Work for one (1) year and, if within one (1) year after the date of Substantial Completion of the Work, any of the Work is found to be not in accordance with the requirements of the Construction Documents, Contractor shall correct it promptly after receipt of written notice from Board to do so, unless Board has previously given Contractor a written acceptance of such condition. Notwithstanding other provisions in the Contract and bid documents all warranties for all the Work shall commence when Substantial Completion is achieved for the entire Project. This obligation under this Subparagraph 12.2.2 shall survive final payment, Acceptance of the Work under the Contract and termination of the Contract and Contractor shall not be relieved or responsibility for negligence, defects of manufacture, faulty materials and/or workmanship within the extent and period herein provided and, upon written notice, Contractor shall remedy any defects due thereto and pay all expenses for any damage to other portions of the Work resulting therefrom.
- 12.2.3 Contractor shall remove from the site portions of the Work which are not in accordance with the requirements of the Construction Documents and are neither corrected by Contractor nor accepted by Board.
- 12.2.4 If Contractor, after notice, fails to promptly correct nonconforming Work in order to comply with the terms of the guarantee, Board may correct it in accordance with Paragraph 2.4 and Contractor and its surety shall be liable for all expenses incurred. Such action by Board shall not relieve Contractor of further guarantee liability.
- 12.2.5 The provisions of Paragraph 12.2 apply to nonconforming Work of the Subcontractors as well as to Work performed directly by Contractor.
- 12.2.6 Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of Board or separate contractors caused by Contractor's correction or removal of Work which is not in accordance with the requirements of the Construction Documents.
- 12.2.7 Contractor specifically warrants and guarantees the integrity of structure provided by the Contractor and/or any elements thereof to be free of water intrusion and/or excessive moisture inside the building as a result of the Contractor's work for a period

of five years.

- 12.2.8.1 Nothing contained in Paragraph 12.2 shall be construed to establish a period of limitation with respect to other obligations which Contractor might have under the Construction Documents. Establishment of the time period of one (1) year as described in Subparagraph 12.2.2 relates only to the specific obligation of Contractor to correct the deficiencies discovered in the Work in that period, and has no relationship to the time within which the obligation to comply with the Construction Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish Contractor's or Surety's liability with respect to Contractor's obligations.
- Unless other provided in the Contract Documents, the Contractor shall continue operate and maintain, at no additional cost to the Board, all systems and equipment related to the Work until the A/E and the Board have declared the Work Substantially Completed, all O&M Manuals have been delivered to the A/E and all training sessions for MDCPS personnel designated by the Board, have been completed as set forth in the Contract Documents. In the event the Contractor fails to provide and perform all items required herein above, Contractor shall continue to be responsible to promptly correct non-conforming Work pursuant to Paragraph 12.2.2 and other requirements of the Contract Documents, whether before or after the expiration of the one (1) year guarantee time period set forth in Paragraph 12.2.2.

12.3 **ACCEPTANCE OF NONCONFORMING WORK**

12.3.1 If Board prefers to accept Work which is not in accordance with the requirements of the Construction Documents, Board may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

ARTICLE 13 - MISCELLANEOUS PROVISIONS

13.1 **GOVERNING LAW AND VENUE**

13.1.1 The Contract shall be governed by the laws of the State of Florida and any applicable Federal laws. Venue for any action brought arising from or related to this contract shall be brought only in a court of competent jurisdiction located in and for Miami-Dade County, Florida.

13.2 SUCCESSORS AND ASSIGNS

13.2.1 Board and Contractor respectfully bind themselves, their successors and legal representatives to the other party hereto and to successors and legal representatives of the other party in respect to covenants, agreements and obligations contained in the Construction Documents. This Contract may not be assigned nor may any assignment of monies due or to become due to Contractor be assigned. If Contractor attempts to make such an assignment, Contractor shall still remain legally responsible for all obligations under the Contract.

13.3 WRITTEN NOTICE

Written notice shall be deemed to have been duly served if delivered in person to the individual or a member of the firm or entity or to an officer of the corporation for which it was intended, or if delivered at or sent by registered or certified mail, regular mail or facsimile to the last business address known to the party giving notice.

13.4 **RIGHTS AND REMEDIES**

- 13.4.1 Duties and obligations imposed by the Construction Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.
- 13.4.2 No action or failure to act by Board, A/E or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed to in writing.

13.5 **TESTS AND INSPECTIONS**

Tests, inspections and approval of portions of the Work required by the Construction Documents or by laws, ordinances, rules, regulations or orders of public authorities having jurisdiction shall be made at an appropriate time. Contractor shall permit and facilitate inspection of the Work by Board, its agents and public authorities having the legal right of entry on the premises at all times. Board's agents include A/E and the Building Code Consultant (BCC). Contractor shall notify A/E and arrange a schedule for the required inspection or test. Board shall provide for the inspection or testing and shall bear the actual cost of performing the test, unless the tests prove unsatisfactory and then Contractor will bear costs associated with testing.

- Work shall not be covered up until Contractor has received notice from A/E that the inspection or test results are satisfactory. If any Work is covered up by Contractor without receipt of notice of satisfactory testing, Contractor shall, if required by A/E, uncover said Work for examination and recover same in a workerlike manner at Contractor's expense.
- 13.5.3 If A/E, Board or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Subparagraph 13.5.1, A/E, upon written authorization from Board, may order such additional testing, inspection or approval. Board shall bear such costs, except as provided in Subparagraph 13.5.4.
- 13.5.4 If such procedures for testing, inspection or approval under Subparagraphs 13.5.1 and 13.5.3 reveal failure of the portions of the Work to comply with requirements established by the Construction Documents, Contractor shall bear all costs made necessary by such failure including those of repeated procedures, costs involved in preparing for the tests, uncovering and recovering related Work, damage to any existing or new Work, and compensation for A/E's services and expenses.
- 13.5.5 Contractor shall bear its own administrative costs related to tests and inspections.
- Tests and inspections conducted pursuant to the Construction Documents shall be made promptly to avoid unreasonable delay in the Work.
- 13.5.7 If mill or factory inspection is called for, Contractor shall notify the suppliers that the material shall not be produced or fabricated without due notice to the inspector employed by Board.
- 13.5.8 The Contractor shall request and obtain all inspections from the Building Official or BCC as required pursuant to the Florida Building Code.
- 13.5.9 The Building Official or BCC may also inspect the work at any other time during construction.
- 13.5.10 The cost of reinspection by the BCC for work previously found to be in noncompliance may be deducted from payments due Contractor.
- 13.6 COMMENCEMENT OF STATUTORY LIMITATION PERIOD
- 13.6.1 As between Board and Contractor:

- .1 Before Substantial Completion. As to acts or failures to act occurring prior to the relevant date of Substantial Completion, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than such date of Substantial Completion;
- .2 Between Substantial Completion and Final Certificate for Payment. As to acts or failures to act occurring subsequent to the relevant date of Substantial Completion and prior to issuance of the Final Certificate for Payment, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than the date of issuance of the final Certificate for Payment; and
- .3 After Final Certificate for Payment. As to acts or failures to act occurring after the relevant date of issuance of the final Certificate for Payment, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than the date of any act or failure to act by Contractor pursuant to any warranty provided pursuant to Paragraph 3.6, the date of any correction of the Work or failure to correct the Work by Contractor under Paragraph 12.2 or the date of actual commission of any other act or failure to perform any duty or obligation by Contractor or Board, whichever occurs last.
- 4. To the extent that applicable Florida Law provides for a longer statute of limitation, the longer time period shall govern the time for bringing action.

13.7 PROVISION OF SITE AMENITIES

- 13.7.1 Contractor shall provide a temporary water line sufficient to supply all water needed for the Work and shall pay for all permits and for all charges for water used by Contractors or Subcontractors.
- 13.7.2 Contractor shall provide sufficient toilet facilities for all workers employed under the Contract.
- 13.7.3 Contractor shall pay for any electrical installation, service or power lines used for the Work. If Contractor desires to use the power metered to Board, permission will be granted provided Contractor connects a power line to the place designated by A/E and pays for the power in accordance with a rate set by Board. Contractor shall give

Board seven (7) days written notice prior to disconnecting.

ARTICLE 14 - TERMINATION OR SUSPENSION OF THE CONTRACT

14.1 **TERMINATION BY CONTRACTOR**

- 14.1.1 Contractor may terminate the Contract if the Work is stopped for a period of thirty (30) days through no act or fault of Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons performing portions of the Work under contract with Contractor for any of the following reasons:
 - .1 issuance of an order of a court or other public authority having jurisdiction;
 - an act of government, such as a declaration of national emergency, making material unavailable;
 - .3 because Board has failed to pay Contractor any payment which has been certified by A/E and authorized by Board within forty-five (45) days after the first regular payment date after certification and approval after notice and subsequent to an additional thirty (30) days after receipt of said notice.
- 14.1.2 If one of the above reasons exists, Contractor may, upon seven (7) additional days' written notice to Board and A/E to cure, terminate the Contract and recover from Board payment for Work executed and for proven loss with respect to materials, equipment, tools and construction equipment and machinery, including reasonable overhead, profit and damages.

14.2 TERMINATION BY BOARD FOR CAUSE

- 14.2.1 Board may terminate the Contract if Contractor:
 - .1 Performs non-conforming, defective or deficient work;
 - .2 Refuses or fails to supply enough properly skilled workers or proper materials;
 - .3 Fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between Contractor and Subcontractor;
 - .4 Disregards laws, ordinances, or rules, regulations or orders of a public

authority having jurisdiction;

- .5 Refuses or fails to correct Work rejected by A/E or failing to conform to the requirements of the Construction Documents;
- .6 Fails to submit schedules pursuant to the provisions of Subparagraph 3.10;
- .7 ails to adhere to the Contract schedule; or
- .8 Otherwise has committedef a material breach of a provision of the Contract or Construction Documents.
- 14.2.2 When any of the above reasons exist, Board, upon certification by A/E that sufficient cause exists to justify such action, or if the position of A/E is vacant, or at its own initiative, Board may without prejudice to any other rights or remedies of Board and after giving Contractor and its surety, if any, seven (7) days' written notice, terminate employment of Contractor and may:
 - .1 Take possession of the Site and of all materials, equipment, tools and construction equipment and machinery thereon owned by Contractor;
 - .2 Accept assignment of subcontracts pursuant to Paragraph 5.4;
 - .3 Finish the Work by whatever reasonable method Board may deem expedient;
 - .4 In lieu of the above the Board may call upon the Surety to perform under the terms and conditions of the Public Performance and Guarantee Bond.
- 14.2.3 When Board terminates the Contract for one of the reasons stated in Subparagraph 14.2.1, Contractor shall not be entitled to receive further payment until the Work is finished.
- 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for A/E's services and expenses made necessary thereby, such excess shall be paid to Contractor. If such costs exceed the unpaid balance, Contractor shall pay the difference to Board. The amount to be paid to Contractor or Board, as the case may be, shall be certified by A/E, upon application, and this obligation for payment shall survive termination of the Contract.
- 14.2.5 Should it be determined by a court of competent jurisdiction that Termination of this

Contract by the Board for Cause was improper, unjustified or unwarranted, the Contractor specifically agrees that such Termination by Board for Cause shall automatically be converted to Termination by Board for Convenience in accordance with the terms and conditions for such set forth in this Contract and Contractor agrees that any claims for additional costs, losses or damages shall be limited to those permitted in this Contract for Termination by Board for Convenience

14.3 SUSPENSION BY BOARD FOR CONVENIENCE

- 14.3.1 Board may, without cause, order Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as Board may determine.
- 14.3.2 An adjustment shall be made in the cost of performance of the Contract, including profit on any increased cost of performance, caused by suspension, delay or interruption. No adjustment shall be made to the extent:
 - .1 that performance is, was or would have been so suspended, delayed or interrupted by another cause for which Contractor is responsible; or
 - .2 that an equitable adjustment is made or denied under another provision of this Contract.
- 14.3.3 Adjustments made in the cost of performance may have a mutually agreed fixed or percentage fee.

14.4. TERMINATION BY BOARD FOR CONVENIENCE

- 14.4.1 The Board may, by written notice to the Contractor, terminate this Contract in whole or in part when it is in the interest of the Board and at the sole discretion of the Board.
- 14.4.2 The Contractor shall be paid for all work executed and reasonable termination expenses and a reasonable allowance for profit and overhead on Work executed, provided that such payments exclusive of termination expenses shall not exceed the total contract price as reduced by other payments made to the Contractor and as further reduced by the value of Work as not completed at the time of termination.
- 14.4.3 The Contractor is not entitled to profit and overhead on work not performed.

ARTICLE 15 - WATER AND SEWER

- The following language is to be used if this project has onsite water and sewer systems and/or offsite water and sewer systems which require an offsite permit by the Utility having jurisdiction or are constructed with the intent of conveying these water and/or sewer facilities to the Utility having jurisdiction.
- The General Contractor has 90 days from award and/or plan approval whichever is later to secure a Public Works permit for offsite work related to water and sewer and paving and drainage, from the Utility having jurisdiction. The General Contractor must commence work on that portion of the project applicable to the offsite permit immediately after securing a Public Works permit or obtain written permission from the owner to reschedule said work. The School Board of Miami-Dade County will not consider any time extension due to the inability of the General Contractor to secure a Public Works permit after plan approval.
- Immediately after the General Contractor has secured said permit and prior to commencing work he shall notify MDCPS Project Manager and arrange a preconstruction meeting with MDCPS (Project Manager, Building Code Compliance Officer and a representative from Offsite Utilities Planning and Development) and the Utilities having jurisdiction and/or governmental agencies having jurisdiction, including but not limited to, Public Works, Miami-Dade Water and Sewer Authority, the Department of Health, Florida Department of Transportation, the Department of Environmental Resource Management and the subcontractor responsible for the work.

At that meeting, the General Contractor shall review the scope of work with, and obtain approval from, the Utilities having jurisdiction and/or governmental agencies having jurisdiction and shall comply with all the requirements, bonds and securities that are imposed by the Utilities having jurisdiction. Once the work has commenced the General Contractor shall coordinate key inspections with the Utilities having jurisdiction, MDCPS, the Engineer and the subcontractor responsible for the work. The General Contractor shall submit a signoff and acceptance letter from the inspector of the Utilities having jurisdiction with his requisition for payment, assuring the Board that the portion of the offsite work which he is submitting payment for has been accepted by the Utilities having jurisdiction. The General Contractor shall secure this letter of acceptance from each governmental agency having jurisdiction performing these inspections. At no time will the Board pay the General Contractor for work that has not passed inspection by the Utilities having jurisdiction.

- Once the offsite work has been completed subject to the above permit the General Contractor shall call for a final inspection from the Utilities having jurisdiction. At this time, the General Contractor shall submit to each Utility having jurisdiction and/or the Board the following documentation which is part of the conveyance package and are the requirements for a final inspection and meter installation by the Utility having jurisdiction.
 - 1. Signed and sealed approved as-builts by a licensed surveyor or engineer of record. As-builts shall comply with the requirements imposed by the Utility having jurisdiction. As-builts submitted to the Miami-Dade Water and Sewer Department (MDWASD) shall comply with Section 01725 "Project As-builts" from the Design and Construction Standard Specifications and Details current edition. This Section includes a requirement to show on the as-built drawings, the Florida State Plane Coordinate (current readjustment) of at least two horizontal control points on or parallel to the centerline of a street within the project boundary. Florida State Plane Coordinate (current readjustment) shall also be shown for all manholes and value boxes constructed, modified or installed as part of the project (some exemptions apply, please see Section 01725 of the MDWASD Construction Standard Specifications and Details). (Submit four (4) blueprints and one (1) set of Mylar to the Utility having jurisdiction and two (2) blue print to the Offsite Utilities Planning and Development).
 - 2. Original DOH Letter of Release with attached bacteriological results. (Submit four (4) originals to the Offsite Utilities Planning and Development).
 - 3. Original Final Construction Report Water. (Submit four (4) originals to the Offsite Utilities Planning and Development).
 - 4. Original DERM Release Letter. (Submit four (4) originals to the Offsite Utilities Planning and Development).
 - 5. Operating and Maintenance manuals for pump stations and onsite sewage collection systems either gravity or force main as required by DERM and prepared by licensed engineer. Including an application for an operating permit for a pump station and the operating permit fee. (Submit four (4) originals to the Offsite Utilities Planning and Development).

Two weeks after the day the Board requests the remainder of the conveyance package the General Contractor shall submit the remainder of the original

conveyance documents to the Board in quadruplicate. These shall include but not be limited to the following:

- Cost breakdown water
- Cost breakdown sewer
- Warranty
- Maintenance Bond
- Waiver and release of lien
- Legal description of easements signed and sealed by a licensed surveyor or engineer of record.
- The Board will not release final payment for offsite or onsite work unless the General Contractor has submitted these conveyance documents.

The General Contractor is responsible for paying each Utility having jurisdiction all construction connection charges and permits associated with the offsite or onsite work.

The General Contractor shall coordinate all offsite water taps for domestic, irrigation and fire line services with the Utility having jurisdiction and shall bear the cost for the installation of these services directly with the Utility having jurisdiction. The General Contractor shall schedule this work with the Utility having jurisdiction in a timely manner. The Board shall not be responsible for the General Contractor's inability to schedule the placement of meters in a timely manner.

The General Contractor shall coordinate all offsite sewer connections with the Utility having jurisdiction and shall bear the cost for the installation of these services directly with the Utility having jurisdiction. The General Contractor shall schedule this work with the Utility having jurisdiction in a timely manner. The Board shall not be responsible for the General Contractor's inability to schedule the sewer connections in a timely manner.

The General Contractor shall be responsible for pulling a permit to abandon or remove all septic tanks according to HRS standards. The General Contractor shall schedule and phase the installation of the waste sewage disposal system in a manner that does not interfere with the school's normal operations or disable the school in any way.

The General Contractor shall secure trenches with chain-link fence and in accordance with the O.S.H.A. Trench Safety Act.

- 15.6 The Building Official will not grant a certificate of occupancy without the permanent water meters for a school facility installed by the Utility having jurisdiction in place and in full compliance with the Utility having jurisdiction's requirements. School Facilities shall not be permitted to open with floating or construction meters. The General Contractor shall bear any and all penalties and fines associated with having an illegal construction or floating meter at the school site at the time of occupancy and shall take all the necessary precautions to secure said floating and construction meter from illegal tampering that can lead to the contamination of the domestic water supply. In case the General Contractor cannot schedule the placement of meters before school opening he shall request in writing to both the Building Official and the Board's Administrative Director of Facilities Operations and Legislative Support or Designee, a request to be able to provide at his expense a supply of potable water for students and staff until such time that permanent water becomes available as well as pay for all fines and penalties resulting from this action. The General Contractor shall also be responsible for securing the domestic water supply in order to protect students and staff. The granting of this request is at the sole discretion of the Building Official and the Board's Administrative Director of Facilities Operations and Legislative Support or Designee.
- 15.7 If the General Contractor through his own fault, has not secured a DERM Release Letter before school opening he shall provide and maintain for the Board temporary portable restrooms at the General Contractor's sole expense in sufficient time for them to be utilized by school opening. These shall remain until the permanent facilities are in use. The number of restrooms provided shall comply with the Florida Building Code and any and all other applicable codes and shall be located in a manner acceptable to and approved by the Board. A water meter for a school facility shall not be installed until the Utility having jurisdiction and MDCPS (Offsite Utilities Planning and Development) receives a DERM Release Letter.

<u>ARTICLE 16 - WATER AND SEWER (CONSTRUCTION MANAGER AT-RISK ONLY)</u>

- The following language is to be used if this project has onsite water and sewer systems and/or offsite water and sewer systems which require an offsite permit by the Utility having jurisdiction or are constructed with the intent of conveying these water and/or sewer facilities to the Utility having jurisdiction.
- The Construction Manager at-Risk has 60 days from the Guaranteed Maximum Price (GMP) approval by The School Board to secure a Public Works Permit for offsite work related to water and sewer and paving and drainage, from the Utility having

jurisdiction. The Construction Manager at-Risk must commence work on that portion of the project applicable to the offsite permit immediately after securing a Public Works permit or obtain written permission from the owner to reschedule said work. The School Board of Miami-Dade County will not consider any time extension due to the inability of the Construction Manager at-Risk to secure a Public Works permit after plan approval.

Immediately after the Construction Manager at-Risk has secured said permit and prior to commencing work he shall notify MDCPS Project Manager and arrange a preconstruction meeting with MDCPS (Project Manager, Building Code Compliance Officer and a representative from Offsite Utilities Planning and Development) and the Utilities having jurisdiction and/or governmental agencies having jurisdiction, including but not limited to, Public works, Miami-Dade Water and Sewer Authority, the Department of Health, Florida Department of Transportation, the Department of Environmental Resource Management and the subcontractor responsible for the work.

At that meeting, the Construction Manager at-Risk shall review the scope of work with, and obtain approval from, the Utilities having jurisdiction and/or governmental agencies having jurisdiction and shall comply with all the requirements, bonds and securities that are imposed by the Utilities having jurisdiction. Once the work has commenced the Construction Manager at-Risk shall coordinate key inspections with the Utilities having jurisdiction, MDCPS, the Engineer and the subcontractor responsible for the work. The Construction Manager at-Risk shall submit a signoff and acceptance letter from the inspector of the Utilities having jurisdiction with his requisition for payment, assuring the Board that the portion of the offsite work which he is submitting payment for has been accepted by the Utilities having jurisdiction. The Construction Manager at-Risk shall secure this letter of acceptance from each governmental agency having jurisdiction performing these inspections. At no time will the Board pay the Construction Manager at-Risk for work that has not passed inspection by the Utilities having jurisdiction.

- Once the offsite work has been completed subject to the above permit the Construction Manager at-Risk shall call for a final inspection from the Utilities having jurisdiction. At this time, the Construction Manager at-Risk shall submit to each Utility having jurisdiction and/or the Board the following documentation which is part of the conveyance package and are the requirements for a final inspection and meter installation by the Utility having jurisdiction.
 - 1. Signed and sealed approved as-builts by a licensed surveyor or engineer of

record. As-builts shall comply with the requirements imposed by the Utility having jurisdiction. As-builts submitted to the Miami-Dade Water and Sewer Department (MDWASD) shall comply with Section 01725 "Project As-builts" from the Design and Construction Standard Specifications and Details current edition. This Section includes a requirement to show on the as-built drawings, the Florida State Plane Coordinate (current readjustment) of at least two horizontal control points on or parallel to the centerline of a street within the project boundary. Florida State Plane Coordinate (current readjustment) shall also be shown for all manholes and value boxes constructed, modified or installed as part of the project (some exemptions apply, please see Section 01725 of the MDWASD Construction Standard Specifications and Details). (Submit four (4) blueprints and one (1) set of Mylar to the Utility having jurisdiction and two (2) blue print to the Offsite Utilities Planning and Development).

- 2. Original DOH Letter of Release with attached bacteriological results. (Submit four (4) originals to the Offsite Utilities Planning and Development).
- 3. Original Final Construction Report Water. (Submit four (4) originals to the Offsite Utilities Planning and Development).
- 4. Original DERM Release Letter. (Submit four (4) originals to the Offsite Utilities Planning and Development).
- 5. Operating and Maintenance manuals for pump stations and onsite sewage collection systems either gravity or force main as required by DERM and prepared by licensed engineer. Including an application for an operating permit for a pump station and the operating permit fee. (Submit four (4) originals to the Offsite Utilities Planning and Development).

Two weeks after the day the Board requests the remainder of the conveyance package the Construction Manager at-Risk shall submit the remainder of the original conveyance documents to the Board in quadruplicate. These shall include but not be limited to the following:

- Cost breakdown water
- Cost breakdown sewer
- Warranty
- Maintenance Bond
- Waiver and release of lien

- Legal description of easements signed and sealed by a licensed surveyor or engineer of record.
- The Board will not release final payment for offsite or onsite work unless the Construction Manager at-Risk has submitted these conveyance documents.

The Construction Manager at-Risk is responsible for paying each Utility having jurisdiction all construction connection charges and permits associated with the offsite or onsite work.

The Construction Manager at-Risk shall coordinate all offsite water taps for domestic, irrigation and fire line services with the Utility having jurisdiction and shall bear the cost for the installation of these services directly with the Utility having jurisdiction. The Construction Manager at-Risk shall schedule this work with the Utility having jurisdiction in a timely manner. The Board shall not be responsible for the Construction Manager at-Risk's inability to schedule the placement of meters in a timely manner.

The Construction Manager at-Risk shall coordinate all offsite sewer connections with the Utility having jurisdiction and shall bear the cost for the installation of these services directly with the Utility having jurisdiction. The Construction Manager at-Risk shall schedule this work with the Utility having jurisdiction in a timely manner. The Board shall not be responsible for the Construction Manager at-Risk's inability to schedule the sewer connections in a timely manner.

The Construction Manager at-Risk shall be responsible for pulling a permit to abandon or remove all septic tanks according to HRS standards. The Construction Manager at-Risk shall schedule and phase the installation of the waste sewage disposal system in a manner that does not interfere with the school's normal operations or disable the school in any way. Construction Manager at-Risk Contractor shall secure trenches with chain-link fence and in accordance with the O.S.H.A. Trench Safety Act.

The Building Official will not grant a certificate of occupancy without the permanent water meters for a school facility installed by the Utility having jurisdiction in place and in full compliance with the Utility having jurisdiction's requirements. School Facilities shall not be permitted to open with floating or construction meters. The Construction Manager at-Risk shall bear any and all penalties and fines associated with having an illegal construction or floating meter at the school site at the time of occupancy and shall take all the necessary precautions to secure said floating and construction meter

from illegal tampering that can lead to the contamination of the domestic water supply. In case the Construction Manager at-Risk cannot schedule the placement of meters before school opening he shall request in writing to both the Building Official and the Board's Administrative Director of Facilities Operations and Legislative Support or Designee, a request to be able to provide at his expense a supply of potable water for students and staff until such time that permanent water becomes available as well as pay for all fines and penalties resulting from this action. The Construction Manager at-Risk shall also be responsible for securing the domestic water supply in order to protect students and staff. The granting of this request is at the sole discretion of the Building Official and the Board's Administrative Director of Facilities Operations and Legislative Support or Designee.

16.7 If the Construction Manager at-Risk through his own fault, has not secured a DERM Release Letter before school opening he shall provide and maintain for the Board temporary portable restrooms at the Construction Manager at-Risk's sole expense in sufficient time for them to be utilized by school opening. These shall remain until the permanent facilities are in use. The number of restrooms provided shall comply with the Florida Building Code and any and all other applicable codes and shall be located in a manner acceptable to and approved by the Board. A water meter for a school facility shall not be installed until the Utility having jurisdiction and MDCPS (Offsite Utilities Planning and Development) receives a DERM Release Letter.

ARTICLE 17 - FORMS

- 17.1 This section contains the latest version of the following forms:
- 17.2 Legal Advertisement for Construction Bids
 Instructions Covering Opening of Bids
 Construction Bid FM-7303
 Trench Safety Act Compliance Statement FM-5238
 Bid Bond FM-3065
 Public Construction Payment Bond FM-5243
 Public Construction Performance & Guarantee Bond FM-5244
 List of Subcontractors FM-3774

Licensure Certification-FM-5384
Lump Sum Construction Contract - FM-7304

Contractor's Requisition for Partial Payment - FM-3071

Final Release of Claim - FM-3070

Required Limits of Insurance DC-INS 02-09